

P/F BANKNORDIK

(Incorporated as a public limited liability company (Fo: partafelag) on the Faroe Islands)

PROSPECTUS REGARDING THE ADMISSION TO TRADING OF DKK 200,000,000 Floating Rate Senior Preferred Notes

This prospectus (the "**Prospectus**") has been prepared by P/F BankNordik, Registration No. 10 with *Skráseting Føroya*, LEI No. 5299005W3ZSR35ZYJH59 ("**BankNordik**" or the "**Issuer**", and together with its direct and indirect subsidiaries and branches, unless otherwise indicated by the context, the "**BankNordik Group**" or the "**Group**"), for the admittance to trading and listing on the regulated market of Nasdaq Copenhagen A/S ("**Nasdaq Copenhagen**") of the DKK 200,000,000 Floating Rate Senior Preferred Notes (the "**Notes**") issued on 22 November 2023 (the "**Issue Date**") in accordance with the terms and conditions of the Notes (see the section "*Terms and Conditions of the Notes*"). An application has been made for the admission of the Notes to trading and listing on Nasdaq Copenhagen. The Issuer expects the first day of trading of the Notes on Nasdaq Copenhagen to be 15 March 2024.

The International Securities Identification Number ("**ISIN**") for the Notes is DK0030529664. The common code for the Notes is 272639264. The Financial Instrument Short Name ("**FISN**") for the Notes is P F BankNordi/6.20/ PrfSnr 2028. Classification of Financial Instruments Code ("**CFI**") for the Notes is DTVUGB.

Unless otherwise defined herein, capitalised terms used in this Prospectus shall have the meaning given to them in the section "*Terms and Conditions of the Notes*" (the "**Terms and Conditions**"). Any reference to a numbered "*Clause*" is to the correspondingly numbered provision of the Terms and Condition.

Unless previously redeemed or purchased and cancelled, the Notes will be redeemed at their Outstanding Principal Amount on 22 November 2028 (the "**Maturity Date**"). The Notes bear interest on their Outstanding Principal Amount at a floating rate from (and including) the Issue Date. The amount of interest payable on each Interest Payment Date (each 22 February, 22 May, 22 August and 22 November) will be determined as the sum of (a) the CIBOR Rate and (b) the Margin.

The Notes constitute Preferred Senior Obligations of the Issuer. As set out in the Terms and Conditions, the Notes constitute direct and unsecured debt obligations of the Issuer and will rank as such as further set out in Clause 7 (*Status of the Notes*) of the Terms and Conditions.

The Notes are issued in uncertificated and dematerialized form through VP Securities A/S ("**Euronext Securities Copenhagen**"). The address of Euronext Securities Copenhagen is Nicolai Eigtveds Gade 8, DK-1402 Copenhagen K, Denmark. Nordea Danmark, Filial af Nordea Bank Abp, Finland performs the tasks of the Paying Agent and Issuing Agent in respect of the Notes and as further set out in the Terms and Conditions. Legal title to the Notes will pass by electronic registration in the book entry system and register maintained by Euronext Securities Copenhagen in accordance with the Danish Capital Markets Act, Executive Orders issued pursuant thereto and the rules and procedures of Euronext Securities Copenhagen from time to time.

The Issuer has been rated A1 (long term issuer rating rating) with a baseline credit assessment of baa1 by Moody's Investor Service (Nordics) AB ("**Moody's**"). For the purposes of Regulation (EC) no. 1060/2009 on credit rating agencies (as amended) (the "**CRA Regulation**"), the credit ratings included or referred to in this Prospectus have been issued by Moody's. Moody's is established in the European Economic Area (the "**EEA**") and is registered under the CRA Regulation. The Notes are unrated. A rating is not a recommendation to buy, sell or hold securities, and may be subject to revision, suspension or withdrawal at any time by the assigning rating agency.

Prospective investors should have regard to the risk factors described under the section "*Risk Factors*" in this Prospectus.

Arranger

Nordea Bank Abp

Dated 14 March 2024

IMPORTANT INFORMATION

This Prospectus is a prospectus in respect of the Notes for the purposes of Article 6 of Regulation (EU) 2017/1129 (the "**Prospectus Regulation**") and for the purposes of giving information with regard to the Issuer and its Group and the Notes which, according to the particular nature of the Issuer and the Notes, is necessary to enable investors to make an informed assessment of the assets and liabilities, financial position, profit and losses and prospects of the Issuer, the rights attaching to the Notes and the reasons for the issuance and its impact on the Issuer. The Issuer has obtained all necessary consent, approvals and authorisations in Denmark in connection with the issue and performance of the Notes. This Prospectus has been approved and registered by the Danish Financial Supervisory Authority (Da: *Finanstilsynet*) (the "**Danish FSA**") pursuant to the Prospectus Regulation and the Commission Delegated Regulation (EU) 2017/1129 (the "**Delegated Prospectus Regulation**") and the Commission Delegated Regulation (EU) 2019/980 supplementing Regulation (EU) 2017/1129 (the "**Delegated Prospectus Regulation**") and the Commission Delegated Regulation (EU) 2019/980 supplementing Regulation (EU) 2017/1129 (the "**Delegated Prospectus Regulation**") and the Commission Delegated Regulation (EU) 2019/980 supplementing Regulation (EU) 2019/979 of 14 March 2019 supplementing the Prospectus Regulation with regard to regulatory technical standards on key financial information in the summary of a prospectus, the publication and classification of prospectuses, advertisements for securities, supplements to a prospectus, and the notification portal, and repealing Commission Delegated Regulation (EU) No 382/2014 and Commission Delegated Regulation (EU) 2016/301, as amended.

The Issuer is responsible for the information in this Prospectus. To the best knowledge of the Issuer, the information contained within this Prospect is in accordance with the facts and does not omit anything likely to affect the import of any information. Where information has been sourced from third parties, this information has been accurately reproduced and, as far as the Issuer is aware and is able to ascertain from the information published by such third parties, no facts have been omitted which would render the reproduced information inaccurate or misleading. The source of third-party information, such as the Faroese Economic Council, is identified where used. This Prospectus is to be read in conjunction with all documents which are deemed incorporated herein by reference (see the section "*Documents Incorporated by Reference*"). Other than in relation to the documents incorporated herein by reference, information on the websites to which this Prospectus refer do not form part of or is deemed incorporated into this Prospectus.

Approval and registration by the Danish FSA do not constitute a guarantee of the information contained in this Prospectus. The Prospectus meets the standards contained in the Prospectus Regulation regarding completeness, comprehensibility and consistency.

To the fullest extent permitted by law, Nordea Bank Abp, with its registered address at Satamaradankatu 5, FI-00020 Nordea, Helsinki, Finland (the "**Arranger**") does not accept any responsibility for the contents of this Prospectus or for any other statement made or purported to be made by the Arranger on its behalf in connection with the Issuer or the issue and admittance to trading and official listing of the Notes. The Arranger accordingly disclaims any and all liability, whether arising in tort or contract or otherwise (save as referred to above), which it might otherwise have in respect of this Prospectus or any such statement.

This Prospectus has been prepared solely for the purpose of admitting the Notes to trading on Nasdaq Copenhagen, a regulated market. This Prospectus does not constitute an offer to sell or the solicitation of an offer to subscribe for or buy any of the Notes in any jurisdiction to any person to whom it would be unlawful to make such an offer in such a jurisdiction. The distribution of this Prospectus may be restricted by law in certain jurisdictions. Accordingly, this Prospectus may not be distributed any jurisdiction where such distribution requires additional prospectus, registration or is contrary to the rules and regulations in such jurisdiction. Persons into whose possession this Prospectus or Notes come must inform themselves about and observe such restrictions. In particular, distribution of this Prospectus and the Notes is restricted in the United States, the EEA and the United Kingdom.

The Notes have not been and will not be registered under the United States Securities Act 1933, as amended, (the "Securities Act"). Subject to certain exceptions, Notes may not be offered, sold or delivered within the United States or to US persons.

The Notes may not be a suitable investment for all investors and each potential investor must determine the suitability of that investment in light of its own circumstances. Particularly each potential investor should (i) have sufficient knowledge and experience to make a meaningful evaluation of the Notes, the merits and risks of investing in the Notes and the information contained or incorporated by reference in this Prospectus or any applicable supplement; (ii) have access to, and knowledge of, appropriate analytical tools to evaluate, in the context of its particular financial situation, an investment in the Notes and the impact such investment will have on its overall investment portfolio; (iii) have sufficient

financial resources and liquidity to bear all of the risks of an investment in the Notes; (iv) understand thoroughly the terms of the Notes and be familiar with the behaviour of any relevant financial markets; and (v) be able to evaluate (either alone or with the help of a financial adviser) possible scenarios for economic, interest rate and other factors that may affect its investment and its ability to bear the applicable risks.

The investment activities of certain investors are subject to legal investment laws and regulations, or review or regulation by certain authorities. Each potential investor should consult its legal advisers to determine whether and to what extent (1) the Notes are legal investments for it, (2) the Notes can be used as collateral for various types of borrowing and (3) other restrictions apply to its purchase or pledge of the Notes. Financial institutions should consult their legal advisers or the appropriate regulators to determine the appropriate treatment of Notes under any applicable risk-based capital or similar rules.

All references in this document to "DKK" refer to Danish Krone and references to "EUR" refer to Euro.

In this Prospectus, unless the contrary intention appears, a reference to a law or a provision of a law is a reference to that law or provision as extended, amended or re-enacted.

This Prospectus has been prepared in English only and is governed by Danish law. The courts of Denmark have jurisdiction over any disputes concerning or related to the contents of this Prospectus.

EU BENCHMARKS REGULATIONS

Amounts payable under the Notes are calculated by reference to CIBOR, which is provided by the Danish Financial Benchmark Facility. As of the date of this Prospectus, the Danish Financial Benchmark Facility does not appear on the register of administrators and benchmarks established and maintained by the European Securities and Markets Authority ("ESMA") pursuant to article 36 of Regulation (EU) 2016/1011 (the "Benchmark Regulation"). As far as the Issuer is aware, the transitional provisions in Article 51 of the Benchmark Regulation apply, such that the Danish Financial Benchmark Facility is not currently required to obtain authorisation or registration.

MIFID II PRODUCT GOVERNANCE/TARGET MARKET

Solely for the purposes of each manufacturer's product approval process, the target market assessment in respect of the Notes has led to the conclusion that: (i) the target market for the Notes is eligible counterparties and professional clients only, each as defined in Directive 2014/65/EU, as amended ("**MiFID II**"); and (ii) all channels for distribution of the Notes to eligible counterparties and professional clients are appropriate. Any person subsequently offering, selling or recommending the Notes (a "**distributor**") should take into consideration the manufacturer's target market assessment; however, a distributor subject to MiFID II is responsible for undertaking its own target market assessment in respect of the Notes (by either adopting or refining the manufacturers' target market assessment) and determining appropriate distribution channels.

IMPORTANT – PROHIBITION OF SALES TO EEA RETAIL INVESTORS

The Notes are not intended to be offered, sold or otherwise made available and should not be offered, sold or otherwise made available to any retail investor in the EEA. For these purposes, a "retail investor" means a person who is one (or more) of: (i) a "retail client", as defined in point (11) of Article 4(1) of MiFID II; (ii) a "customer" within the meaning of Directive (EU) 2016/97 (the "**Insurance Distribution Directive**"), where that customer would not qualify as a "professional client" as defined in point (10) of Article 4(1) of MiFID II; or (iii) not a "qualified investor" as defined in the Prospectus Regulation. Consequently, no key information document required by Regulation (EU) No. 1286/2014 (the "**EU PRIIPs Regulation**") for offering of selling the Notes or otherwise making them available to retail investors in the EEAA has been prepared and therefore offering or selling the Notes or otherwise making them available to any retail investor in the EEA may be unlawful under the EU PRIIPs Regulation.

IMPORTANT - PURSUING LIQUIDATION OR BANKRUPTCY CLAIMS AGAINST THE ISSUER IN DENMARK

The Issuer is incorporated on the Faroe Islands. If proceedings with respect to the liquidation or bankruptcy of the Issuer should occur, the Noteholders would be required to pursue their claims on the Notes in proceedings with respect to the Issuer on the Faroe Islands. To the extent that the relevant Noteholders are entitled to any recovery with respect to such Notes in any such Faroese bankruptcy proceedings, such Noteholders would be entitled to a recovery in Danish Kroner.

Table of Contents

IMPORT	IMPORTANT INFORMATION		
1.	Responsibility Statement	5	
2.	Risk Factors	6	
3.	Documents incorporated by reference	24	
4.	Terms and Conditions	25	
5.	Amount and Use of Proceeds	50	
6.	Description of BankNordik and the BankNordik Group	51	
7.	General Information	69	

1. Responsibility Statement

1.1 The Issuer's Responsibility

P/F BankNordik is responsible for this Prospectus in accordance with Danish law.

1.2 Statement

We hereby declare as the persons responsible for this Prospectus on behalf of BankNordik in our capacity as members of the Board of Directors and the Executive Management of BankNordik that, to the best of our knowledge, the information contained in this Prospectus is in accordance with the facts and that this Prospectus makes no omission likely to affect its import.

We furthermore declare that this Prospectus has been approved by the Danish FSA as competent authority under the Prospectus Regulation. The Danish FSA only approves this Prospectus as meeting standards of completeness, comprehensibility and consistency imposed by the Prospectus Regulation. Such approval should not be considered as an endorsement of BankNordik that is the subject of this Prospectus or be considered as an endorsement of the quality of the Notes that are the subject of this Prospectus. Prospective investors should make their own assessment as to the suitability of investing in the Notes.

P/F BankNordik, 14 March 2024

1.3 Board of Directors

1.4

Birita Sandberg Samuelsen	Rúni Vang Poulsen	Tom Ahrenst
Chairman	Deputy chairman	Board member
Birger Durhuus	Alexandur Johansen	Kenneth Samuelsen
Board member	Board member	Board member
Rúna Hentze	Marjun Eystberg	Kristian Reinert Davidsen
Board member	Board member	Board member
Executive Management		

Turið F. Arge Chief Executive Officer

2. Risk Factors

In purchasing Notes, investors assume the risk that the Issuer may become insolvent or otherwise be unable to make or restricted from making all payments due in respect of the Notes. There is a wide range of factors which individually or together could result in the Issuer becoming unable to make or restricted from making all payments due in respect of the Notes. The Issuer may not be aware of all relevant factors and certain factors which it currently deems not to be material may become material as a result of the occurrence of events outside the Issuer's control. The Issuer has identified in this Prospectus a number of factors which could materially adversely affect its business and ability to make payments due under the Notes.

The material risks, as currently assessed by the Issuer, taking into account (i) the expected magnitude of their negative impact on the Issuer and/or the Notes and (ii) the probability of their occurrence, are set out first in the risk factor categories *"Risk factors that may affect the Issuer's ability to fulfil its obligations under the Notes"* and *"Risk factors relating to the structure of the Notes"*. Where it has been assessed not to be possible to make an accurate assessment of the probability of the occurrence of a risk factor in these risk factor categories, the probability of the occurrence of such risk factor has not been specified in respect of such risk factor.

In addition, factors which are material for the purpose of assessing the market risks associated with the Notes are also described below.

Prospective investors should also read the detailed information set out elsewhere in this Prospectus and reach their own views prior to making any investment decision.

Words and expressions defined in the Terms and Conditions or elsewhere in this Prospectus have the same meanings in this section, unless otherwise stated. References to a numbered "Clause" shall be to the relevant Clause in the Terms and Conditions.

2.1 Risk factors that may affect the Issuer's ability to fulfil its obligations under the Notes

2.1.1 Risks related to the general economic and geopolitical conditions on the Faroe Islands and internationally

The business activities and performance of Group are dependent on the level of banking, finance and financial services required by its customers. In particular, levels of borrowing are dependent on customer confidence, employment trends, state of the economy, housing market and market interest rates at the time. As the Group currently conducts the majority of its business on the Faroe Islands, where its market share on the Faroe lending market is approximately 40 per cent and 38 per cent on deposits, its performance is influenced by the level and cyclical nature of business activity on the Faroe Islands. This is in turn affected by both domestic and international economic and political events, including military actions, geopolitical conflicts, instability on global commodities and financial markets and epidemics and pandemics. Given the size and relatively non-diversified (compared to larger economies) economy of the Faroe Islands, macroeconomic trends and events that impact one sector of the Faroese economy may indirectly have an adverse impact on the broader Faroese economy and in turn the Group's business.

For example, Russia's invasion of Ukraine (and the related sanctions imposed by the United States, the EU and the United Kingdom, among others) has and is expected to continue to affect the global economy in an adverse manner and thereby also directly and indirectly the Faroese economy. The Issuer has no direct or indirect business in either Ukraine or Russia, however, in general a large part of the total export from the Faroe Islands has in previous years been to Russia in the form of various fish products and Faroese fishing vessels have had some access to fisheries in Russian territorial

waters. Russia's actions in Ukraine, have led to and may lead to new severe sanctions against Russia and Russia have responded and may respond with sanctions towards e.g., the Faroe Islands. This may over time negatively impact one or more of the Issuer's corporate customers ability to honour their fiscal obligations towards the Issuer due to either lower export sales and/or no or restricted access to fisheries in Russian territorial waters. As i) the main export to the Russian market is and has been fish products which also may be exported to other markets outside Russia, e.g. United States, EU, the United Kingdom and certain African countries, and ii) restricted access to fisheries in Russian territorial waters may free up quotas of Russian vessels in Faroese territorial waters the negative economic effect of sanctions on Faroese exports may be partly mitigated by directing such products to such other markets and the utilisation by Faroese vessels of quota previously granted Russian vessels in Faroese waters.

The Group has no significant business in the United Kingdom but the Group's fish product exporting customers and other business customers with exposure to or a trading relationship with the United Kingdom may be adversely affected by the new trading relationship between with the United Kingdom under the EU-UK Trade and Cooperation Agreement.

The recent escalation of conflict in the Middle East with the outbreak of the Israel-Hamas war may also have wide reaching negative effects on global markets which could have a material adverse effect on the Group's business, results of operations and financial position. Currently, however, it is not possible to make any predictions about the broader implications of this.

In 2022, inflation rates increased and also became more broad-based as prices for most categories of consumer spending increased. Inflation rates have continued to increase on the Faroe Islands during parts of 2023, albeit not on the same level as in 2022. Inflation rates in 2024 have been lower than at the same time in both 2023 and 2022, however, they are still fluctuating and it is difficult to foresee how they will develop for the remainder of 2024. The Danish National Bank increased its deposit rate in increments throughout 2022 and 2023 from (negative) (0.5) per cent to 3.6 per cent p.a. in September 2023, which indirectly has affected the Faroe Islands. The issuer has largely followed the Danish National Bank's increase of deposit rates by raising both the deposit and lending rates. Rising interest rates have also caused the costs of external funding to rise.

The Group is also exposed to adverse effects and macroeconomic implications of an outbreak of a pandemic or epidemic, similar to the COVID-19 pandemic. COVID-19 lockdowns and combatting measures had an adverse effect on many of the industries and sectors towards which the Group is exposed, including fisheries, manufacturing, retail, tourism, entertainment and service industries, such as hotels and restaurants. As the Group is one of relatively few larger financial institutions that operate in the Faroe Islands, and considering the significant position of the aforementioned sectors in the Faroese economy, many of the Group's customers were directly affected by the COVID-19 pandemic and mitigating measures. This also affected the Issuer which in 2020 suspended its financial guidance for that year due to the COVID-19 pandemic. Should a similar event occur in the future, the Group's clients operating in such industries may become unable to repay their loans, resulting in increased default rates and increased credit impairments. This could adversely affect the revenue and profits of the Group.

A negative development in the general economic conditions on the Faroe Islands, such as a downturn in the economy, an increase in unemployment on the Faroe Islands, strikes and lockouts as a result of collective bargaining, reduced personal income levels of the Group's customers (for example due to higher interest rates, surge in food prices and increase in electricity (and other utilities) prices), inflationary pressures, borrowers' reduced ability to repay loans, disruptions in global supply chains, or a reduction in the value of housing and other collateral provided to the Group could have a material adverse effect on the Group's business, results of operations, financial position and/or prospects. Ultimately, the combined or individual effect of such events and could cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes, potentially resulting in a depreciation in the market price of the Notes and investors losing all or part of their investment.

2.1.2 Credit risk related to borrowers, counterparties and customers of the Group

Credit risk is the risk of loss that the Group may incur as a result of borrowers or other counterparties of the Group defaulting on their payment obligations, including the risks attaching to the Group's customers having financial difficulties, risks relating to large exposures and concentration risks that may occur in relation to the Group's business and risks attaching to granted, unutilised credit lines that may be provided by the Group. Credit risk is also the risk that the Group may be unable to assess the credit risk of potential borrowers or other counterparties and may provide loans and advances to customers that increase the Group's credit risk exposure more than intended. Credit risk is an inherent part of the Group's business. Ordinary credit risk arises from the Issuer's loan portfolio and from credit lines and guarantees. The loan portfolio consists of 45 per cent to the private sector, 47 per cent to the corporate sector and 8 per cent to the public sector. The Group has a strong loan-tovalue for housing loans. 92 per cent of total housing loans are below a loan-to-value of 80 per cent and 65 per cent of total housing loans are below 40 per cent of loan-to-value. Furthermore, credit risk also includes settlement and counterparty risks. Settlement risk is the risk arising when payments are settled, for instance payments for currency transactions and trading in financial instruments, including derivatives. The risk arises when the Issuer transfers payments before it has attained full assurance that the counterparty has met all its obligations. Counterparty risk is the risk of loss as a result of a customer's default of OTC derivatives and securities financing instruments. Credit risk also arises from credit investments in the Group in, for example, senior bonds of other highly rated financial institutions. Market-related counterparty credit risk arises from financial instruments including fixed income, equity and other investments that the Group owns or is in another way exposed to.

The Group's total risk-weighted items amounts to DKK 6,818,861,000 as of 31 December 2023. Of this, the risk-weighted items with credit risk amounts to DKK 5,808,267,000 as of 31 December 2023, the risk-weighted items with market risk amounts to DKK 347,722,000 and the risk-weighted items with operational risk amounts to DKK 622,873,000 as of 31 December 2023.

The Group may suffer losses from credit risk in the future, some of which may be material in amount, which could have an adverse effect on the Group's business, results of operations, financial position and/or prospects. As a consequence thereof, the Issuer may become unable to pay interest, principal or other amounts on or in connection with the Notes, potentially resulting in a depreciation in the market price of the Notes and investors losing all or part of their investment.

2.1.3 Market risk related to adverse developments in market values resulting from fluctuations in interest rates, credit spreads, foreign currency exchange rates and equity and commodity prices

The Group faces market risks as an inherent part of its business. The Group deals and takes positions in financial products such as interest-based products, shares and foreign exchange instruments, which expose the Group to various market-based risks, including the risk of fluctuations in interest rates, credit spreads, exchange rates and equity and commodity prices. If the Group is unable to adequately foresee and take the necessary actions to adapt to such fluctuations, the materialisation of these events may result in reduced revenue and/or profits of the Group. For example, interest rates risk attaching to positions in the trading book of the Group derives primarily from bonds and swaps. In a recession or in times of uncertainty on the financial markets, the interest rates and prices of such instruments may fluctuate significantly over short periods of time. With respect to the banking book, interest rate risk of the Group derives from fixed-rate deposits and

lending from ordinary banking transactions, monetary policy loans, bonds as well as interest rate risk related to the Group's own funding. The performance of financial markets may cause changes in the value of the Group's investment and trading portfolios as well as affect other areas of the operations of the Group such as the availability of funding for the Group. A significant part of the Group's exposure to the risks associated with the financial markets derives from changes in the value of its bond portfolio which accounts for approximately DKK 1.4 billion. The bond portfolio is mostly fixed-rated inconvertible bonds allocated to approximately 55 per cent government bonds and 45 per cent mortgage bonds. The book value of this portfolio has over the past few years been subject to relatively significant fluctuations.

Any fluctuations in interest rates, foreign currency exchange rates, equity prices and fixed income prices which the Group is unable to account for in a timely manner could have an adverse effect on the Group's business, results of operations, financial position and/or prospects. In turn, these direct negative impacts may cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes, potentially resulting in a depreciation in the market price of the Notes and investors losing all or part of their investment.

2.1.4 Funding and liquidity risk related to funding costs, liquidity and refinancing risk, deposit withdrawal and access to funds

Liquidity risk is the risk of loss that the Group may incur because funding costs become excessive, a lack of funding prevents Group from fulfilling its business model or a lack of funding prevents the Group from fulfilling its payment obligations. Refinancing risk is the risk of the Group not being able to refinance maturing deposits, senior debt, covered bonds or other liabilities, or the risk that the refinancing cost will be so high that it will adversely affect net interest income of the Group.

For the Group, being a financial intermediary, liquidity and refinancing risk is an inherent and unavoidable part of the Group's banking operations. Liquidity and refinancing risk of the Group arises from funding mismatches in the balance sheet as the average duration of the Issuer's loan portfolio is generally longer than the average duration of the Issuer's funding sources.

As a retail bank, the Issuer receives a high portion of its funding from customer deposits, and therefore the Group is also subject to the risk that its depositors could withdraw their funds at a faster rate than the rate at which the Group's borrowers repay their loans, thus causing liquidity strains for the Group.

Ready access to funds is essential to any banking business, including the Group. If the Group is unable to access funds or to access the markets from which the Group raises funds, it could have an adverse effect on the Group's ability to meet its obligations as they fall due and impede the Group's ability to finance its operations adequately. These and other factors could also lead creditors to form a negative view of the Group's liquidity, which could result in higher borrowing costs and decreased access to various funding sources for the Group which could have an adverse effect on Group's business, results of operations, financial position and/or prospects and could thereby ultimately cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes.

2.1.5 Risks related to an increase in the Issuer's and/or the Group's capital requirements, leverage ratio requirements, net stable funding ratio requirement, liquidity requirements and/or risk exposure amount (REA)

The Issuer and the Group are subject to relatively complex regulation relating to capital, solvency, leverage ratio, liquidity and other requirements. Compliance with these regulations is generally a

source of administrative costs as well as lost potential revenue (e.g. due to capital being retained for buffer purposes instead of being lent out/invested).

Minimum and additional own funds requirement

The regulatory capital management framework applicable to the Issuer is provided by the CRR/CRD. Under this framework, the Issuer is required to hold a minimum amount of regulatory capital equal to 8 per cent of the risk exposure amount ("**REA**") (of which at least 4.5 per cent must be Common Equity Tier 1 Capital, and at least 6 per cent must be Tier 1 capital) of the Issuer, which is also referred to as the minimum own funds Pillar 1 requirements (the "minimum own funds requirements"). In addition to Pillar 1 competent authorities may require additional "Pillar 2" capital to be maintained by an institution (the "additional own funds requirements") or the "individual solvency requirement").

The REA of the Issuer and the Group and the capital requirements (the minimum own funds requirements and the additional own funds requirements the combined buffer requirement (see "Combined capital buffer requirement" below) applicable to the Issuer and/or the Group are, by their nature, calculated by reference to a number of factors any one of which or combination of which may not be easily observable or capable of calculation by investors. Any of the minimum own funds Pillar 1 requirements, additional own funds requirements or buffer capital requirements applicable to the Issuer and/or the Group may be amended in the future to include new and more onerous capital requirements, which may exacerbate the risk that discretionary payments, including payments of interest on the Notes, are cancelled, which is likely to have an adverse effect on the price of the Notes and the likelihood of investors getting a return on their investment.

Combined capital buffer requirement

CRD includes a combined capital buffer requirement consisting of a capital conservation buffer, an institution-specific countercyclical capital buffer, a G-SII buffer (applicable to global systemically important institutions ("G-SIIs"), a O-SII buffer (applicable to other systemically important institutions ("O-SIIs") and a systemic risk buffer.

At the date of this this Prospectus BankNordik's combined capital buffer requirement consists of a 2.5 per cent capital conservation buffer, a 2.0 per cent O-SII buffer (also referred to as systemically important financial institution ("SIFI") buffer), a 2 per cent systemic risk buffer and a countercyclical buffer, currently at 0 per cent (all are stated as a percentage of the overall risk exposure, except for the SIFI buffer and systemic risk buffer which are stated as a percentage of the Faroese exposures, and all must be met with Common Equity Tier 1 capital). The volatility in the financial markets over the past years have resulted in various adjustments to systemic risk and countercyclical buffers. At the date of this Prospectus, it is not possible to predict the future development of the systemic risk buffer requirements are likely to result in additional costs for Group, which may adversely affect the Group's results of operations.

CRR Amendment Regulation, CRD Amendment Directive and SREP Guidelines

Regulation (EU) 2019/876 (the "**CRR Amendment Regulation**") (as implemented on the Faroe Islands by act no. 648 of 18 May 2015 as amended by act no. 883 of 21 June 2022 and most recently amended by the Order no. 561 of 11 May 2023) and Directive (EU) 2019/878 (the "**CRD Amendment Directive**") (as partly implemented on the Faroe Islands by decree no. 1589 of 19 December 2017,

as amended by decree no. 1032 of 29 June 2022 and most recently amended by the decree no. 547 of 23 May 2023) impose various leverage requirements on the Issuer and the Group.

Such requirements include a leverage ratio requirement of 3 per cent Tier 1 Capital, a leverage ratio related maximum distributable amount for G-SIIs (the "L-MDA"), harmonised binding requirement for stable funding (the "**Net Stable Funding Ratio**" or "**NSFR**") on 100 per cent, strengthening of the conditions for use of internal models and changes to the relevant regulator's application of the institution specific "Pillar 2" capital add-ons (referred to above as the additional own funds requirements). The additional own funds requirement must be fulfilled with at least 56.25 per cent Common Equity Tier 1 Capital and at least 75 per cent Tier 1 capital.

Furthermore, the CRD Amendment Directive authorises the relevant competent authority to require that a regulated institution fulfils its additional own funds requirement with a higher portion of Tier 1 Capital or Common Equity Tier 1 Capital where necessary (while having regard to the specific circumstances of the relevant institution). The CRD Amendment Directive also introduces a so-called "guidance on additional own funds" requirement, which sets a level and quality of CET1 capital the relevant credit institution is expected to hold in excess of its overall capital requirement. The guidance which is expected to become applicable to the Group on additional own funds will be based on, inter alia, the stress tests performed in respect of the Issuer and the Group (see the risk factor "*Risks related to stress tests and other regulatory enquiries, which could trigger enforcement actions by supervisory authorities may have a material adverse effect on the Group's business, results of operations, financial position and/or prospects" in section 2.1.7 below). Repeated failure to meet the guidance on additional own funds may entitle the competent authority to take supervisory measures and impose additional own funds requirements, potentially resulting in increased administrative costs and decreased revenue.*

A G-SII that fails to meet its applicable leverage ratio buffer requirement shall calculate its L-MDA. The L-MDA will, among other things, set the level for payments on Additional Tier 1 Capital instruments (such as the Notes). According to the CRR Amendment Regulation, the European Commission was required to submit a report to the European Parliament and to the Council on whether it is appropriate to extend the L-MDA restrictions to O-SIIs, such as the Issuer. On 16 February 2021, the European Commission published that report, which concluded that the European Commission does not consider it appropriate to introduce a leverage ratio surcharge for O-SIIs in the current context. Instead, the European Commission proposed that this question should be examined as part of the comprehensive review of the macroprudential toolbox in banking by 30 June 2022, as set out in Article 513 of the CRR. As of the date of this Prospectus, no legislative proposal on extending the L-MDA restrictions to O-SIIs has been published.

The CRR Amendment Regulation and CRD Amendment Directive entered into force on 27 June 2019 and have varying application dates, ranging from 12 months to four years after entry into force. On 22 December 2020 Act no. 2110 on Changes to the Financial Business Act, the Recovery and Resolution Act of Certain Financial Undertakings, the Capital Markets Act and Cessation of the Act on Finansiel Stabilitet (changes as result of the revision of the Capital Requirements Directive (CRD V) and the Resolution and Recovery Directive (BRRD II) etc.) (the "**BRRDII/CRDV Act**") was adopted by the Danish Parliament. The BRRDII/CRDV Act implements the CRD Amendment Directive and directive (EU) 2019/879 (the "**BRRD Amendment Directive**") (see the risk factor "*Resolution tools and powers under the BRRD*" – "*The BRRD Amendment Directive*" in section 2.2.2) into Danish law. Parts of, *inter alia*, the BRRDII/CRDV Act have been implemented on the Faroe Islands by decree no. 547 of 23 May 2023. The Faroese implementation of the CRD Amendment Directive. The implementation on the Faroe Islands includes provisions on MREL Requirements and on the competent resolution authority on the Faroe Islands. At the date of this Prospectus, it remains uncertain whether the CRR Amendment Regulation and CRD Amendment Directive will impose

additional capital, liquidity, and/or leverage requirements on the Issuer and/or the Group, potentially affecting the Issuer's ability to perform its obligations under the Notes.

The European Banking Authority ("**EBA**") and the Danish Financial Supervisory Authority (the "**Danish FSA**") will continue to propose detailed rules through binding technical standards, guidelines, recommendations and/or opinions in respect of many areas, including the CRR, the CRR Amendment Regulation, the CRD and the CRD Amendment Directive. As a consequence, the Group is subject to the risk of possible interpretational changes. Given the uncertainty of the exact wording of the technical standards, they could potentially lead to a reduction in the regulatory capital or an increase in the REA of the Issuer and the Group, which in turn may affect the Issuer's ability to fulfil its obligations under the Notes.

On 7 December 2017, the Basel Committee published its recommendations named Basel III: Finalising post crisis reforms (informally referred to as "Basel IV"). The reforms contain new requirements for credit risk, operational risk, CVA risk and a so-called output floor which sets new minimum standards for capital requirements in financial institutions using internal models for calculating capital requirements. On 27 October 2021, the European Commission published its proposal for a review of the CRR Regulation and the CRD Directive, implementing, inter alia, the Basel IV (the "Basel IV CRR/CRD Proposal"). The Basel IV CRR/CRD Proposal is now subject to the EU legislative procedure. On 24 January 2023, the European Parliament's Economic and Monetary Affairs Committee (ECON) announced the adoption of draft reports on the Basel IV CRR/CRD Proposal. On 27 June 2023 negotiators from the European Council and the European Parliament reached a provisional agreement on the Basel IV CRR/CRD Proposal. The Basel IV CRR/CRD Proposal introduces, inter alia, an output floor on 72.5 per cent of the standardised approach for calculating REA. The output floor will be gradually introduced from 1 January 2025 over a period of 5 years. The Group's REA will increase as a result of the implementation of Basel IV as set out in the Basel IV CRR/CRD Proposal. The exact amount with which the REA of the Group will increase is currently unknown.

Pursuant to the Danish FSA's decision of 30 November 2023 (published by the Danish FSA) setting the minimum requirement for own funds and eligible liabilities for the Issuer, the Issuer is required to maintain a minimum requirement of own funds and eligible liabilities (MREL) (Da: *nedskrivningsegnede passiver*) corresponding to 21.7 per cent of REA applicable from 1 June 2024 and which will gradually increase to 25.5 per cent of REA on 1 June 2026.

The relationship between any of the aforementioned or future incremental additional own funds requirements and the combined buffer requirement is complex and depends on various factors. If the Issuer's capital levels fall below the combined buffer requirement, it may lead to severe adverse consequences for the Issuer. For example, if the regulatory capital requirements, leverage ratio requirements, liquidity restrictions or ratios applied to the Issuer and/or the Group are increased in the future, any failure of the Issuer and/or the Group to maintain such increased capital and liquidity ratios could result in administrative actions or sanctions, which could have a material adverse effect on the Group's business, results of operations, financial position and/or prospects and could thereby ultimately cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes, which is likely to cause the price of the Notes to depreciate.

Faroese and Greenlandic real estate markets

On 29 November 2023 the Danish Financial Supervisory Authority published a memorandum on "Loss levels on exposures secured by mortgages on real estate" where the Danish FSA assessed whether the real estate markets in Denmark, the Faroe Islands and Greenland are well-developed and long-established.

For the Faroe Islands and Greenland in particular the memorandum concludes that based on the established criteria the Danish FSA has not been able to determine whether there is a well-developed and long-established market for residential and commercial properties. The assessment significantly affects financial institutions' risk-weighting of exposures secured by mortgages on residential and commercial property.

Therefore, the Danish FSA has subsequently asked financial institutions operating in the North Atlantic including BankNordik for further information and clarification on the Faroese and Greenlandic property markets.

On this basis financial institutions are providing supplementary material to the Danish FSA based on their deep local knowledge of the Faroese and Greenlandic real estate markets to support an assessment by the Danish FSA that the markets are well-developed and long-established.

The Danish FSA is expected to reach its conclusion on the status of the Faroese and Greenlandic property markets during the first half of 2024.

The Group has noted that as a result of the consultation the Danish FSA may increase the riskweighting on the Faroese and Greenlandic banks' exposures secured by mortgages on residential and commercial property. This in turn may necessitate an increased capital base to maintain solvency levels as well as meeting the Supervisory Diamond limit values.

If such possible changes are implemented the Issuer expects to manage its loan portfolio with respect to its largest exposures within some months such that it in the future can comply with its dividend policy.

As a result of the risk of an increased risk-weighting of exposures secured by mortgages on residential and commercial property the board of directors has proposed a dividend of 26 per cent of net profit on the annual general meeting in 2024, thus deviating from its dividend policy of 70 per cent.

2.1.6 **Risks related to the operations, business and reputation of the Group**

Operational risk is understood as the risk of loss that the Group may incur which results from inefficient or deficient internal procedures in the Group, from human or systemic errors or from external events, including legal risks. Model risk, which is the risk of loss that the Group may incur as a consequence of decisions based mainly on output from internal models and occurring due to errors in the development, implementation or use of such models, is also defined as operational risk.

All activities of the Group are subject to operational risk. Operational risks are categorised on the basis of the seven event types defined by Basel III: Employment practices and workplace safety; external fraud; business disruption and systems failures; internal fraud; clients, products and business practice; execution/delivery and process management; and damage to physical assets. Moreover, the Group is exposed to the risk of cybercrime attacks.

The Group's business and other activities (including those performed by the Issuer), are increasingly dependent on highly advanced IT systems. The Group may be the target of malicious hacking resulting in the shutdown of individual or all of its IT systems. The consequences hereof could include financial losses, business disruption, inability to service payments on time, loss of data or other sensitive information etc.

Business risk is the risk of loss that the Group may incur caused by changes in external circumstances or events that harm the Group's image or operational performance. Business risk includes strategic risk and reputational risk. Strategic risk and reputational risk are the risks of loss

that the Group may incur due to external circumstances or events that could harm the Issuer's reputation or affect its earnings negatively.

Failure by the Group to identify and manage these risks could have a material adverse effect on the Group's business, results of operations, financial position and/or prospects and could thereby ultimately cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes.

2.1.7 **Risks related to stress tests and other regulatory enquiries, which could trigger enforcement** actions by supervisory authorities

The Issuer is subject to periodic stress testing and other regulatory enquiries to examine the resilience of banks to adverse market developments. Such stress tests are initiated and coordinated by the Danish Central Bank and/or the national supervisors such as the Danish FSA and/or the Faroese systemic risk Council. Stress tests and the disclosure of their results by supervisory authorities can influence the banking or the financial services sector and lead to a loss of trust with regard to individual banks or the financial services sector as a whole.

The Group's own liquidity stress testing model may prove inadequate, which could lead to a nonsatisfactory outcome of regulatory stress tests applied on the Group. While this has not occurred before, the potential material adverse effect of a non-satisfactory outcome of stress tests may be far reaching. For example, the Group could suffer reputational harm, increased funding costs, become subject to increased regulatory scrutiny and/or enforcement actions by supervisory authorities, as well as the impose of higher capital and liquidity requirements of the Group. As discussed in the preceding risk factors, each of these circumstances could have an adverse effect on the Group's results of operation, business, revenues and financial position, as well as the Issuer's ability to perform its obligations in respect of the Notes.

2.1.8 Risks relating to the Group becoming involved in supervisory actions, litigation and regulatory investigations

The Group operates in a legal and regulatory environment that exposes it to potentially significant litigation and regulatory risks. As a result, the Group is and may become involved in various disputes and legal proceedings on the Faroe Islands and other jurisdictions, including litigation and regulatory investigations. Further, the Group's banking and other operations, like those of other credit institutions, have been the subject of regulatory scrutiny from time to time. For example, the Group is subject to applicable anti-money laundering and terrorist financing laws. The Danish FSA conducts on-going inspections from time to time of the Group's compliance with anti-money laundering and terrorist financing laws, which can potentially lead to supervisory actions. Such supervisory actions and any other disputes and legal proceedings are subject to many uncertainties, and their outcomes are often difficult to predict, particularly in the earlier stages of a case or investigation. Therefore, if the Group become involved in supervisory actions, litigation and regulatory investigations it could have an adverse effect on Group's business, results of operations, financial position and/or prospects and could thereby ultimately cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes.

2.1.9 Risks relating to the Group's participation in the Deposit Guarantee Scheme and resolution fund may result in the Group to incur additional costs

On the Faroe Islands and other jurisdictions, deposit guarantee schemes and similar funds (each, a "**Deposit Guarantee Scheme**") have been implemented from which compensation for deposits may become payable to customers of financial services firms in the event that such financial services firm is unable to pay, or unlikely to pay, claims against it. Revised legislation regarding the Danish Deposit

Guarantee Scheme redefines the Danish scheme as a premium based scheme funded by the banking sector itself, such that the participating banks' (including the Issuer) payments into the scheme will be more stable every year in profit and loss terms. The calculation of premium will be based on each participating bank's covered deposits and the relevant bank's risk profile. The premium payments will stop when a target level of 0.8 per cent of covered deposits has been reached. In addition, the Issuer contributes to the Danish resolution fund established as the Danish resolution financing arrangement under the BRRD, which capital must amount to 1.0 per cent of the covered deposits of all Danish credit institutions by 31 December 2024. The future target level of funds to be accumulated in Deposit Guarantee Schemes and resolution funds across different EU countries may exceed the minimum levels provided for in the BRRD, Directive 2014/49/EU, as amended from time to time (the "Revised Deposit Guarantee Schemes Directive"), and in EU Regulation No 806/2014, as amended from time to time and EU Regulation No 81/2015, as amended from time to time of the European Parliament and of the Council establishing uniform rules and a uniform procedure for the resolution of credit institutions and certain investment firms in the framework of a Single Resolution Mechanism and a Single Resolution Fund (the latter of which will be relevant should Denmark choose to participate in the Single Resolution Mechanism). Both the BRRD and the Revised Deposit Guarantee Schemes Directive are implemented in Danish law as referred to in the risk factor "Resolution tools and powers under the BRRD" in section 2.2.2 below and by Consolidated Act no. 356 of 2 April 2020 on Depositor and Investor Guarantee Scheme as amended from time to time. Adjustments to the relevant Faroese regulation relating to solvency and liquidity requirements of financial businesses to implement, inter alia, the Revised Deposit Guarantee Schemes Directive have been made by executive order no. 561 of 11 May 2023.

On 18 April 2023, the European Commission adopted a proposal to adjust and further strengthen the existing EU bank crisis management and deposit insurance (CMDI) framework (the "CMDI Proposal"). The CMDI Proposal includes, *inter alia*, certain amendments to the Revised Deposit Guarantee Schemes Directive, including in relation to the coverage of public entities as well as temporary high balances on bank accounts (see risk factor "*Resolution tools and powers under the BRRD*" – "*Depositor preference and the general bail-in tool*" in section 2.2.2 below). It is still unclear whether Denmark will join the European Banking Union and therefore be a part of the Single Resolution Mechanism. It therefore remains unclear whether the Issuer, being resident in a country outside the EU, can participate under the Single Resolution Mechanism and the Single Resolution Fund or similar schemes and if so the level of costs the Issuer will incur in the coming year in relation to payments to deposit guarantee funds and/or resolution funds on a national or European level. The CMDI Proposal has not been implemented into neither Danish nor Faroese law as of the date of this Prospectus.

2.2 Risk factors relating to the structure of the Notes

Set out below are various risk factors which relate to the Notes and the structure of the Notes, including their designation as "Senior Preferred Debt".

2.2.1 If the CMDI Proposal is adopted and implemented into Faroese law, the Notes would rank junior to all depositors of the Issuer

The Notes constitute direct, unsubordinated and unsecured debt obligations of the Issuer and rank as described in Clause 7 (*Status of the Notes*) of the Terms and Conditions. The Notes will rank at least *pari passu* with all outstanding senior unsecured and unsubordinated obligations of the Issuer (save for obligations which may be preferred by law, including obligations benefitting from a preferred ranking to the Notes), present and future. Currently, the Notes also rank *pari passu* with depositors of the Issuer (other than in respect of preferred and covered deposits).

If the CMDI Proposal is adopted as proposed (and implemented on the Faroe Islands), one element of the CMDI Proposal would mean that the Notes will no longer rank *pari passu* with certain deposits of the Issuer; instead, the Notes would rank junior in right of payment to the claims of all depositors. As such, there may be an increased risk of an investor in the Notes losing all or some of their investment. The CMDI Proposal, if adopted and implemented on the Faroe Islands, may also lead to a rating downgrade for the Notes. See "*Risk factors related to the market which may affect the Issuer and/or the Notes – Credit rating risks related to the Issuer and/or the Notes*" for further information on credit ratings.

Additionally, the Issuer may issue other unsubordinated obligations or instruments that rank or are expressed to rank senior to the Notes as regards the right to receive periodic payments on a liquidation or bankruptcy of the Issuer and the right to receive repayment of capital on a liquidation or bankruptcy of the Issuer. In the event of a liquidation or bankruptcy of the Issuer. In the event of a liquidation or bankruptcy of the Issuer will be required to pay all its obligations that rank senior to the Notes in full before it can make any payments on the Notes. If this occurs, the Issuer may not have enough assets remaining after these payments are made to pay amounts due under the Notes.

In addition, in the event of a liquidation or bankruptcy of the Issuer, to the extent the Issuer has assets remaining after paying its creditors who rank senior to the Notes, payments relating to other obligations or instruments of the Issuer that rank *pari passu* with the Notes may, if there are insufficient assets to satisfy the claims of all of the Issuer's *pari passu* creditors, further reduce the assets available to pay amounts due under the Notes on a liquidation or bankruptcy of the Issuer

2.2.2 **Resolution tools and powers under the BRRD**

Recovery and Resolution Directive

The BRRD, including the general bail-in tool and the minimum requirement for own funds and eligible liabilities ("**MREL**"), has been implemented into Danish law and entered into force in Denmark on 1 June 2015 by the Danish Act on Recovery and Resolution of certain Financial Businesses (and set in force on the Faroe Islands pursuant to the decree no. 1589 of 19 December 2017 and most recently amended by the decree no. 1032 of 21 June 2022 as amended or replaced from time to time) and by amendments to the Danish Financial Business Act as in force on the Faroe Islands.

Any reference to the BRRD below shall include the implementation hereof into Faroese law.

The BRRD confers substantial powers on national resolution authorities and is designated by Member States with a credible set of tools to intervene sufficiently early and quickly in relation to unsound or failing credit institutions, investment firms, certain financial institutions and certain holding companies (each, a "**relevant entity**") to ensure the continuity of the relevant entity's critical financial and economic functions while minimizing the impact of a relevant entity's failure on the economy and financial system.

The BRRD contains various resolution powers which may be used alone or in combination where the relevant resolution authority considers that (a) a relevant entity is failing or likely to fail, (b) there is no reasonable prospect that any alternative private sector measures would prevent the failure of such relevant entity within a reasonable timeframe, and (c) a resolution action is in the public interest. A relevant entity will be considered as failing or likely to fail when either: (i) it is, or is likely in the near future to be, in breach of its requirements for continuing authorisation; (ii) its assets are, or are likely in the near future to be, less than its liabilities; (iii) it is, or is likely in the near future to be, unable to pay its debts as they fall due; or (iv) it requires extraordinary public financial support (except in limited circumstances). In such circumstances, the relevant resolution authority may use the following resolution tools and powers without the consent of the relevant entity's creditors: (i) sale of business – which enables resolution authorities to direct the sale of the relevant entity or the whole or part of

its business on commercial terms; (ii) bridge institution – which enables resolution authorities to transfer all or part of the business of the relevant entity to a "bridge institution" (an entity created for this purpose that is wholly or partially in public control), which may limit the capacity of the relevant entity to meet its repayment obligations; (iii) asset separation – which enables resolution authorities to transfer assets (including, without limitation, impaired or problem assets) to one or more publicly owned asset management vehicles to allow them to be managed with a view to maximizing their value through eventual sale or orderly wind-down (this can be used together with another resolution tool only); and (iv) bail-in – which gives resolution authorities the power to write down certain claims of unsecured creditors of a failing relevant entity (which write-down may result in the reduction of such claims to zero) and to convert certain unsecured debt claims (including the Notes) to equity or other instruments of ownership (the "general bail-in tool"), which equity or other instruments of ownership could also be subject to any future application of the general bail-in tool.

If the relevant resolution authorities implement, or indicate the implementation of, resolution measures or other powers vested in them by the BRRD vis-á-vis the Group, it may have an adverse impact on the rights of the Noteholders and the Issuer's ability to perform its obligations under the Notes.

Depositor preference and the general bail-in tool

As part of the reforms required by the BRRD, amendments have been made to relevant legislation in Denmark (and put into force on the Faroe Islands pursuant to degree 1584 of 21 December 2017) to establish a preference in the insolvency hierarchy for certain deposits that are eligible for protection by the Danish deposit guarantee scheme and the uninsured element of such deposits and, in certain circumstances, deposits made in non-EEA branches. In addition, the Danish implementation of the Revised Deposit Guarantee Scheme increased the nature and quantum of insured deposits to cover a wide range of deposits, including certain corporate deposits (unless the depositor is a public sector body or financial institution) and some temporary high value deposits. The effect of these changes is to increase the size of the class of preferred creditors. All such preferred deposits will rank in the insolvency hierarchy ahead of all other unsecured creditors of the Issuer, including the Noteholders. Furthermore, insured deposits are excluded from the scope of the general bail-in tool. As a result, if the general bail-in tool were exercised by the relevant resolution authority, the Notes would be more likely to be bailed-in than certain other unsubordinated liabilities of the Issuer such as other preferred deposits.

It is noted that the CMDI Proposal looks to amend the BRRD, including, among other things, the amendment of the ranking of claims in insolvency to provide for a general depositor preference, pursuant to which the insolvency laws of EU Member States, including Denmark, would be required by the BRRD to extend the legal preference of claims in respect of deposits relative to ordinary unsecured claims to all deposits. The implementation of the CMDI Proposal is subject to further legislative procedures but, if adopted and subsequently implemented on the Faroe Islands in its current form, this would mean that the Notes will rank junior to the claims of all depositors, including deposits of large corporates and other deposits that are currently excluded from the above privileged claims.

Any such general depositor preference would also impact upon any application of the general bail-in tool, as such application is to be carried out in the order of the hierarchy of claims in normal insolvency proceedings. Accordingly, this would mean that following any such amendment of the insolvency laws of Denmark (and to the extent implemented on the Faroese Islands) to establish a general depositor preference, any resulting write-down or conversion of the Notes by the relevant Danish resolution authority would be carried out before any write-down or conversion of the claims of depositors, such as those of large corporates that previously would have been written-down or converted alongside such Notes. One of the stated objectives of this proposed amendment is to

reduce the likelihood of deposits generally needing to be included in any such write-down or conversion upon any application of the general bail-in tool and improve the process for the application of the general bail-in tool. However, this may have the corresponding impact of increasing the likelihood of any write-down or conversion of the Notes. The above will only be applicable in respect of the Issuer if the CMDI Proposal is implemented through an EU regulation and by a decree on the Faroe Islands (in each case in its current form).

If the CDMI Proposal is implemented on the Faroe Islands the relative ranking of the Notes to deposits will shift in in favour of deposits and if write-down powers and/or the general bail-in tool are exercised in respect of the Issuer, the Notes may depreciate in value to a larger extent than they otherwise would without the implementation of the CDMI Proposal.

MREL requirement

With the implementation of the BRRD, European banks are required to have bail in-able resources in order to fulfil their MREL Requirement. There is no minimum EU-wide level of the MREL Requirement – each resolution authority is required to make a separate determination of the appropriate MREL Requirement for each resolution group within its jurisdiction, depending on the resolvability, risk profile, systemic importance and other characteristics of each institution.

The Danish FSA on 30 November 2023 set the MREL Requirement for the Issuer at 25.5 per cent of the Issuer's REA to be met by 1 June 2026. The requirement will be phased in as follows:

1 June 2024: 21.7 per cent of the Issuer's REA (6.0 per cent of the Issuer's total exposure).

1 June 2025: 23.6 per cent of the Issuer's REA (6.0 per cent of the Issuer's total exposure).

1 June 2026: 25.5 per cent of the Issuer's REA (6.0 per cent of the Issuer's total exposure).

The Danish FSA on 30 November 2023 also set a requirement for subordination of the Issuer's own funds and eligible liabilities at 27.8 per cent of the Issuer's total liabilities and own funds to be met by 1 June 2026. The requirement will be phased in as follows:

1 June 2024: 24.1 per cent of the Issuer's REA.

1 June 2025: 26.0 per cent of the Issuer's REA.

1 June 2026: 27.8 per cent of the Issuer's REA.

According to the Danish FSA's MREL decision regarding the Issuer, the MREL Requirement is no longer determined based on the issuer's total liabilities and own funds but as a percentage of the REA and the total exposure, respectively.

The MREL Requirement for each relevant entity will be set annually as a percentage of:

- (i) the total risk exposure amount of the undertaking calculated in accordance with Article 92(3) of the CRR, and
- (ii) the total exposure measure of the undertaking calculated in accordance with Articles 429 and 429a of the CRR.

Accordingly, an increase in the total capital requirement (e.g., an increase in the countercyclical capital buffer) and/or an increase in the exposures of the relevant entity could lead to a higher MREL requirement.

Exercise of powers under the BRRD

The BRRD outlines the priority ranking of certain deposits in an insolvency hierarchy. The BRRD establishes a preference in the ordinary insolvency hierarchy, firstly for insured depositors and, secondly, for all other deposits of individuals and micro, small and medium-sized enterprises held in the EEA or non-EEA branches of an EEA bank. These preferred deposits rank ahead of all other unsecured senior creditors of the Issuer in the insolvency hierarchy. Furthermore, the insolvency hierarchy could be changed in the future.

Any application of the general bail-in tool under the BRRD shall be in accordance with the hierarchy of claims in normal insolvency proceedings. Accordingly, the impact of such application on Noteholders will depend on their ranking in accordance with such hierarchy, including any priority given to other creditors such as depositors.

To the extent any resulting treatment of holders of Notes pursuant to the exercise of the general bailin tool is less favourable than would have been the case under such hierarchy in normal insolvency proceedings, a holder has a right to compensation under the BRRD based on an independent valuation of the relevant entity (which is referred to as the "no creditor worse off principle" under the BRRD). Any such compensation is unlikely to compensate that holder for the losses it has actually incurred and there is likely to be a considerable delay in the recovery of such compensation.

The exercise of any power under the BRRD or any suggestion of such exercise, could have a material adverse effect on the rights of Noteholders, the price or value of their investment in the Notes and/or the ability of the Issuer to satisfy its obligations under the Notes. Although the BRRD contains certain limited safeguards for creditors in specific circumstances, including in the case of senior creditors a safeguard that aims to ensure that they do not incur greater losses than they would have incurred had the relevant entity been wound up under normal insolvency proceedings, that the Issuer is not in a position to state whether these safeguards will be effective if such powers are exercised. The determination that any power under the BRRD shall be exercised or that all or a part of the principal amount of the Notes will be subject to bail-in is likely to be inherently unpredictable and depends on a number of factors which are outside of the Issuer's control. The application of the general bail-in tool with respect to the Notes may result in the write-down or cancellation of all, or a portion of, the principal amount of, or outstanding amount payable in respect of, and/or interest on, the Notes and/or the conversion of all, or a portion, of the principal amount of, or outstanding amount payable in respect of, or interest on, the Notes into shares or other securities or other obligations of the Issuer or another person, including by means of a variation to the terms of the Notes to give effect to such application of the general bail-in tool. Accordingly, potential investors in the Notes should consider the risk that the general bail-in tool may be applied in such a manner as to result in Noteholders losing all or a part of the value of their investment in the Notes or receiving a different security than the Notes, which may be worth significantly less than the Notes and which may have significantly fewer protections than those typically afforded to debt securities. Moreover, the relevant resolution authority may exercise its authority to apply the general bail-in tool without providing any advance notice to the Noteholders. The exercise of any power under the BRRD or any suggestion of such exercise could, therefore, materially adversely affect the rights of the relevant Noteholders, the price or value of their investment in the Notes and/or the ability of the Issuer to satisfy its obligations under the Notes.

2.2.3 The regulation and reform of "benchmarks" may adversely affect the value of Notes linked to or referencing such "benchmarks"

Interest rates and indices which are deemed to be "benchmarks" (such as, in the case of the Notes a reference rate, CIBOR, are the subject of recent national and international regulatory guidance and proposals for reform. Some of these reforms are already effective whilst others are still to be

implemented. These reforms may cause such benchmarks to perform differently than in the past, to disappear entirely, or have other consequences which cannot be predicted. Any such consequence could have a material adverse effect on any Notes linked to or referencing such a benchmark.

The so-called "Benchmarks Regulation" (Regulation (EU) 2016/1011 of the European Parliament and of the Council of 8 June 2016 on indexes used as reference values for financial instruments and financial agreements or for measuring investment fund results and amending Directives 2008/48/EC and 2014/17/EU and Regulation (EU) No 596/2014) applies, subject to certain transitional provisions to the provision of benchmarks, the contribution of input data to a benchmark and the use of a benchmark within the EU. Among other things, it (i) requires benchmark administrators to be authorised or registered (or, if non-EU based, to be subject to an equivalent regime or otherwise recognised or endorsed) and (ii) prevents certain uses by EU supervised entities (such as the Issuer) of benchmarks of administrators that are not authorised or registered (or, if non-EU based, not deemed equivalent or recognised or endorsed).

The Benchmarks Regulation could have a material Impact on any Notes linked to or referencing a benchmark, in particular, if the methodology or other terms of the benchmark are changed in order to comply with the requirements of the Benchmarks Regulation. Such changes could, among other things, have the effect of reducing, increasing or otherwise affecting the volatility of the published rate or level of the benchmark, in turn affecting the interest payable in respect of the Notes.

Currently, work is undertaken in Denmark and internationally to reform and/or find alternatives to existing (in full or in part) estimate-based reference rates in most currencies, with a view to securing the use of transaction-based reference rates to a greater extent. In the long term, this may be of importance to the development and use of Danish reference rates, such as CIBOR.

In November 2021, Danmarks Nationalbank assumed responsibility for DESTR (Denmark short-term rate). Danmarks Nationalbank began publishing DESTR on 4 April 2022. The first publication reflected trading activity on 1 April 2022. Thus, DESTR has been available for use in financial contracts with effect from 1 April 2022. The impact of DESTR on CIBOR is currently unclear.

It is not possible to predict with certainty whether, and to what extent, CIBOR and other benchmarks will continue to be supported going forwards. This may cause CIBOR and other benchmarks to perform differently than they have done in the past and may have other consequences which cannot be predicted.

More broadly, any of the international or national reforms, or the general increased regulatory scrutiny of benchmarks, could increase the costs and risks of administering or otherwise participating in the setting of a benchmark and complying with any such regulations or requirements. Such factors may have the following effects on certain benchmarks: (i) discourage market participants from continuing to administer or contribute to the benchmark; (ii) trigger changes in the rules or methodologies used in the benchmark or (iii) lead to the disappearance of the benchmark.

Investors should consult their own independent advisers and make their own assessment about the potential risks imposed by the Benchmarks Regulation reforms in making any investment decision with respect to any Notes linked to or referencing a "benchmark".

2.2.4 Further discontinuance of certain benchmark rates (for example, CIBOR) may adversely affect the value of Notes which are linked to or which reference any such benchmark rate

Investors should be aware that, if a benchmark rate such as CIBOR were discontinued or otherwise unavailable, the rate of interest on the Notes which are linked to or which reference such benchmark rate will be determined for the relevant period by the fallback provisions applicable to such Notes. The Terms and Conditions provide for certain fallback arrangements in the event that a published

benchmark, such as CIBOR, (including any page on which such benchmark may be published (or any successor service)) becomes unavailable or if a CIBOR Rate Event (as defined in the Terms and Conditions below) otherwise occurs.

If the circumstances described in the preceding paragraph occur and the CIBOR Rate Replacement is specified in the relevant Clauses as being applicable, such fallback arrangements will include the possibility that:

- the relevant rate of interest (or, as applicable, component thereof) could be set or, as the case may be, determined by reference to a Successor CIBOR Rate or an Alternative CIBOR Rate (as applicable) determined by the Issuer (following consultation with an Independent Advisor (if any)); and
- (ii) such Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) may be adjusted (if required) by the Issuer (following consultation with an Independent Advisor (if any)) in order to reduce or eliminate, to the extent reasonably practicable in the circumstances, any economic prejudice or benefit (as applicable) to investors arising out of the replacement of the CIBOR Rate with the relevant Successor CIBOR Rate or Alternative CIBOR Rate (as applicable),

in any such case, acting in good faith an in a commercially reasonable manner as described more fully in the Terms and Conditions.

In addition, the Issuer (following consultation with an Independent Adviser (if any)) may also determine (acting in good faith and in a commercially reasonable manner) that other amendments to the Terms and Conditions of the Notes are necessary in order to follow market practice in relation to the relevant Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) and to ensure the proper operation of the relevant Successor CIBOR Rate or Alternative CIBOR Rate (as applicable).

No consent of the Noteholders shall be required in connection with effecting any relevant Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) or any other related adjustments and/or amendments described above.

If, following the occurrence of a CIBOR Rate Event, no Successor CIBOR Rate or Alternative CIBOR Rate is determined, the ultimate fallback for determining the rate of interest for a particular Interest Period may result in the rate of interest for the last preceding Interest Period being used. This may result in the effective application of a fixed rate for the Notes based on the rate which was last observed on the relevant screen page for the purposes of determining the rate of interest in respect of an Interest Period. In addition, due to the uncertainty concerning the availability of Successor CIBOR Rates and Alternative CIBOR Rates, the relevant fallback provisions may not operate as intended at the relevant time.

Any such consequences could have a material adverse effect on the value of and return on any such Notes. Moreover, any of the above matters or any other significant change to the setting or existence of any relevant rate could affect the ability of the Issuer to meet its obligations under the relevant Notes or could have a material adverse effect on the value or liquidity of, and the amount payable under, the relevant Notes. Investors should note that, in the case of relevant Notes, the Issuer will have discretion to adjust the relevant Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) in the circumstances described above. Any such adjustment could have unexpected commercial consequences which, depending on the particular circumstances of each Noteholder, may or may not be favourable to each Noteholder.

In addition, potential investors should also note that:

- (i) no Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) will be adopted, and no other amendments to the Terms and Conditions of the Notes will be made, if and to the extent that, in the determination of the Issuer, the same could reasonably be expected to prejudice the eligibility of the Notes for the purposes of the debt buffer requirement of the Issuer and/or the MREL Requirement (if applicable) of the Issuer; and/or
- (ii) no Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) will be adopted, and no other amendments to the terms and conditions of the Notes will be made, if and to the extent that, in the determination of the Issuer, the same could reasonably be expected to result in the Relevant Regulator treating the next Interest Payment Date as the effective maturity of the Notes, rather than the Maturity Date.

2.3 Risks related to the market which may affect the Issuer and/or the Notes

Set out below is a brief description of certain market risks, including credit rating risk, liquidity risk, exchange rate risk and interest rate risk.

2.3.1 Credit rating risks related to the Issuer

One or more independent credit rating agencies may assign credit ratings to the Issuer. At the date of this Prospectus, the Issuer has been rated A1 (long term issuer rating) with a baseline credit assessment of baa1 by Moody's. The Issuer's credit ratings are important to its business and can help attract new investors. In addition, the a "good" rating may enable the Issuer to obtain better terms when raising future capital. Any relevant rating agency, including Moody's, may downgrade the ratings of the Issuer or the ratings of the Issuer's debt instruments, including the Notes (however, the Notes are unrated as of the date of this Prospectus) either as a result of the financial position of the Group or changes to applicable rating methodologies used by any relevant rating agency. Substantial changes in the Issuer's credit ratings of the Notes, potentially at a considerable cost of capital, which may have an adverse impact on the Group's business and the Group's ability to pay amounts due under the Notes issued under this Prospectus.

A rating agency's evaluation of the Issuer may also be based on a number of factors not entirely within the control of the Issuer, such as conditions affecting the financial services industry generally. Any reduction in the Issuer's credit ratings or the ratings of its debt instruments, including any unsolicited credit rating, could adversely affect its liquidity and competitive position, undermine confidence in the Issuer and the Group, increase its borrowing costs, limit its access to the capital markets, or limit the range of counterparties willing to enter into transactions with the Issuer and the Group's business, financial situation, results of operations, liquidity and/or prospects and could thereby ultimately cause the Issuer to become unable to pay interest, principal or other amounts on or in connection with the Notes.

The credit rating of the Notes and/or the Issuer or the Group could decline after the date of this Prospectus. This may result in depreciation of the market value of the Notes, causing investors to lose all or part of their investment.

2.3.2 The secondary market of the Notes

The Notes may have no established trading market and one may never develop. If an active trading market does not develop or is not maintained, the market price and liquidity of the Notes may be

adversely affected. If a market does develop, it may not be very liquid and any liquidity in such market could be significantly affected by any purchase and cancellation of the Notes by the Issuer. In total, 200 Notes with a nominal value of DKK 1 million each have been issued which will be admitted to trading and official listing on Nasdaq Copenhagen. Accordingly, the volume and the features of the Notes are in the Issuer's opinion not likely to facilitate a liquid market for the Notes.

Therefore, investors may not be able to sell their Notes easily or at prices that will provide them with a yield comparable to similar investments that have a developed secondary market. It should further be noted that the volatility and liquidity of a capital instrument compared to that of other capital instruments will be affected by a number of factors, for example, the size and nature of the issuer, the currency of the instrument, the interest rate, the ranking of the capital instrument and whether capital instrument has been structured to meet the investment requirements of limited categories of investors. Illiquidity may have an adverse effect on the market value of the Notes.

2.3.3 Exchange rate risks and exchange controls

The Issuer will pay principal and interest on the Notes in the DKK. This presents certain risks relating to currency conversions if an investor's financial activities are denominated principally in a currency or currency unit (the "**Investor's Currency**") other than the DKK. These include the risk that exchange rates may significantly change (including changes due to devaluation of the Specified Currency or revaluation of the Investor's Currency) and the risk that authorities with jurisdiction over the Investor's Currency may impose or modify exchange controls. An appreciation in the value of the Investor's Currency relative to the DKK would decrease (1) the Investor's Currency equivalent yield on the Notes, (2) the Investor's Currency equivalent value of the principal payable on the Notes and (3) the Investor's Currency equivalent market value of the Notes.

Government and monetary authorities may impose (as some have done in the past) exchange controls that could adversely affect an applicable exchange rate or the ability of the Issuer to make payments in respect of the Notes. As a result, investors may receive less interest or principal than expected, or no interest or principal, reducing their return on their investment in the Notes.

3. Documents incorporated by reference

The following documents which have previously been published or are published simultaneously with this Prospectus shall be incorporated in, and form part of, this Prospectus:

- (i) the sections "Statement and reports" and "Financial statement" of the audited consolidated annual report of the BankNordik Group for the financial year ended 31 December 2023 which are available on the Issuer's website: <u>https://www.banknordik.com/reports/financial-reports</u> and this <u>direct link</u> covering pages 27 through 107 of the 2023 annual report,
- (ii) the sections "Statement and report" and "Financial statement" of the audited consolidated annual report of the BankNordik Group for the financial year ended 31 December 2022 which are available on the Issuer's website: <u>https://www.banknordik.com/reports/financial-reports_and this_direct_link_covering</u> pages 28 through 108 of the 2022 annual report,
- (iii) the BankNordik Group's Risk Management Report for 2023 which is available on the Issuer's website: https://www.banknordik.com/reports/risk-management-reports.

Such documents shall be incorporated in, and form part of, this Prospectus, save that any statement contained in a document which is incorporated by reference herein shall be modified or superseded for the purpose of this Prospectus to the extent that a statement contained herein modifies or supersedes such earlier statement (whether expressly, by implication or otherwise). Any statement so modified or superseded shall not, except as so modified or superseded, constitute a part of this Prospectus. Any documents themselves incorporated by reference in the documents incorporated by reference in this Prospectus shall not form part of this Prospectus.

Those parts of the 2023 annual report and the 2022 annual report which are not specifically incorporated by reference into this Prospectus, are either not relevant for investors in the Notes or are covered elsewhere in this Prospectus.

4. Terms and Conditions

Date of Terms and Conditions	20 November 2023
Issuer:	P/F BankNordik
Registration No.:	10 (Skráseting Føroya)
LEI No.:	5299005W3ZSR35ZYJH59
ISIN:	DK0030529664
Issuing Agent:	Nordea Danmark, Filial af Nordea Bank Abp, Finland
Paying Agent:	Nordea Danmark, Filial af Nordea Bank Abp, Finland
Calculation Agent:	Nordea Danmark, Filial af Nordea Bank Abp, Finland

1. Overview of terms of Notes

Notes	Floating Rate Preferred Senior Notes due 22 November 2028		
Initial Amount:	DKK 200,000,000	Danish kroner Two hundred million	
Denomination:	DKK 1,000,000		
Minimum settlement amount:	DKK 1,000,000		
Currency:	DKK		
Issue Date:	22 November 2023		
Maturity Date:	22 November 2028		
Redemption Price:	Outstanding Principal Amounts, together with accrued interest (if any) thereon insofar as it has not been cancelled		
First Call Date:	22 November 2027	Outstanding Principal Amounts, together with accrued interest (if any) thereon	
Call:	Interest Payment Date Tax Event MREL Disqualification Event (subject to approval, cf. Clause 11.8 (Conditions to redemption etc. prior to the Maturity Date))	Outstanding Principal Amounts, together with accrued interest (if any) thereon	
Interest accrual:	From Issue Date		
Interest Rate:	CIBOR Rate (3 months) <u>plus</u> Margin		
Margin:	2.25 per cent per annum		
Interest Payment Date:	Quarterly on 22 February, 22 May, 22 August and 22 November, first time on 22 February 2024		
Day count fraction:	Actual/360, Adjusted		
Business Day Convention:	Modified Following Business Day		

2. Introduction

- 2.1 These terms and conditions dated 20 November 2023 (the "Terms and Conditions") relate to the DKK 200,000,000 Floating Rate Preferred Senior Notes (the "Notes") to be issued by P/F BankNordik (Business Registration (*Skráseting Føroya*) No. 10) incorporated on the Faroe Islands with its registered address at Oknarvegur 5, FO-110 Tórshavn, Faroe Islands and its telephone number is +298 330330 (the "Issuer").
- 2.2 The Issuer is a public limited liability company (Fo: *partafelag*) incorporated under the laws of Faroe Island. The Issuer is a credit institution (Da: *pengeinstitut*) under supervision by the Danish Financial Supervisory Authority (Da: *Finanstilsynet*).
- 2.3 The issue of the Notes was authorised and approved by the Issuer's board of directors at a meeting held on 25 September 2023.
- 2.4 The Notes will be issued on 22 November 2023 (the "**Issue Date**") at an issue price of 100.00 per cent.
- 2.5 Nordea Danmark, Filial af Nordea Bank Abp, Finland acting as issuing agent (the "**Issuing Agent**") will perform the task of registering the Notes in the book entry system of Euronext Securities Copenhagen.
- 2.6 Nordea Danmark, Filial af Nordea Bank Abp, Finland (the "**Paying Agent**") will perform the task of arranging for payment of any amount due under the Notes through Euronext Securities Copenhagen (subject to in each case having received the relevant amount from the Issuer) in accordance with these Terms and Conditions.
- 2.7 Nordea Danmark, filial af Nordea Bank Abp, Finland in its capacity as calculation agent (the "**Calculation Agent**") will perform the task of calculating any rate of interest and any amount, including any interest amounts, due under the Notes in accordance with these Terms and Conditions.

3. Definitions

In addition to the terms defined above the following expressions have the following meanings in these Terms and Conditions:

"Additional Amounts" has the meaning ascribed to it in Clause 12 (Taxation (Gross up)).

"Adjustment Spread" means a spread (which may be positive, negative or zero), formula or methodology for calculating a spread, which the Independent Adviser (in consultation with the Issuer) determines is required to be applied to a Successor CIBOR Rate or an Alternative CIBOR Rate in order to reduce or eliminate, to the extent reasonably practicable in the circumstances, any economic prejudice or benefit to the Noteholders as a result of the replacement of CIBOR Rate with a Successor CIBOR Rate or an Alternative Old or methodology which:

(a) in the case of a Successor CIBOR Rate, is formally recommended in relation to the replacement of the applicable CIBOR Rate with the relevant Successor CIBOR Rate by any Relevant Nominating Body;

- (b) in the case of a Successor CIBOR Rate for which no recommendation has been made or in the case of an Alternative CIBOR Rate, the Independent Adviser (in consultation with the Issuer) determines is recognised or acknowledged as being in customary usage in Danish and/or international debt capital markets transactions which reference the applicable CIBOR Rate, where such rate has been replaced by the relevant Successor CIBOR Rate or Alternative CIBOR Rate; or
- (c) if no such customary market usage is recognised or acknowledged, the Independent Adviser (in consultation with the Issuer) in its discretion, determines (acting in good faith) to be appropriate.

"Alternative CIBOR Rate" means the rate that the Independent Adviser determines has replaced the CIBOR Rate in customary market usage in the relevant debt capital markets for the purposes of determining rates of interest in respect of notes denominated in Danish Kroner and of a comparable duration to the relevant Interest Period, or, if the Independent Adviser determines that there is no such rate, such other rate as the Independent Adviser determines in its sole discretion is most comparable to the CIBOR Rate.

"Applicable MREL Regulations" means at any time the laws, regulations, requirements, guidelines and policies then in effect on the Faroe Islands giving effect to any MREL Requirement or any successor regulations then applicable to the Issuer and/or the Group, including, without limitation to the generality of the foregoing, CRD/CRR, the BRRD and those regulations, requirements, guidelines and policies giving effect to any MREL Requirements or any successor regulations, and then in effect (whether or not such requirements, guidelines or policies have the force of law and whether or not they are applied generally or specifically to the Issuer and/or the Group).

"BRRD" means the Directive (2014/59/EU) of the European Parliament and of the Council on resolution and recovery of credit institutions and investment firms dated 15 May 2014 and published in the Official Journal of the European Union on 12 June 2014 (or, as the case may be, any provision of Danish or Faroese law transposing or implementing such Directive), as amended or replaced from time to time (including, for the avoidance of doubt, the amendments to such Directive resulting from Directive (EU) 2017/2399 of the European Parliament and of the Council dated 12 December 2017 2019 and published in the Official Journal of the European Union on 27 December 2017 and Directive (EU) 2019/879 of the European Parliament and of the Council dated 10 Directive (EU) 2019/879 of the European Union on 7 June 2019).

"**Business Day**" shall mean a day on which banks and Euronext Securities Copenhagen are open for general business (including dealing in foreign exchange and foreign currency deposits) in Copenhagen.

"Calculation Agent" has the meaning ascribed to it in Clause 2.7.

"Call Date" means the First Call Date and any Interest Payment Date falling thereafter.

"CIBOR Rate" has the meaning ascribed to it in Clause 8.1.3(a) or, following the occurrence of a CIBOR Rate Event, the Successor CIBOR Rate or Alternative CIBOR Rate replacing the CIBOR Rate in accordance with Clause 9.

"CIBOR Rate Amendments" has the meaning ascribed to it in Clause 9.3.2.

"CIBOR Rate Determination Date" has the meaning ascribed to it in Clause 9.2.1(a).

"CIBOR Rate Event" has the meaning ascribed to it in Clause 9.2.1

"CRD/CRR" means, as the context requires, any or any combination of the CRD Directive, the CRR and any CRD/CRR Implementing Measures.

"CRD Directive" means the Directive (2013/36/EU) of the European Parliament and of the Council on prudential requirements for credit institutions and investment firms dated 26 June 2013 and published in the Official Journal of the European Union on 27 June 2013 (or, as the case may be, any provision of Danish or Faroese law transposing or implementing such Directive), as amended or replaced from time to time (including, for the avoidance of doubt, the amendments to such Directive resulting from Directive (EU) 2019/878 of the European Parliament and of the Council as regards exempted entities, financial holding companies, mixed financial holding companies, remuneration, supervisory measures and powers and capital conservation measures dated 20 May 2019 and published in the Official Journal of the European Union on 7 June 2019).

"CRD/CRR Implementing Measures" means any regulatory capital rules or regulations, or other requirements, which are applicable to the Issuer and/or the Group, as applicable, and which prescribe (alone or in conjunction with any other rules, regulations or other requirements) the requirements to be fulfilled by financial instruments for their inclusion in the regulatory capital of the Issuer (on a non-consolidated or consolidated basis) to the extent required by the CRD Directive or the CRR, including for the avoidance of doubt and without limitation any regulatory technical standards, guidelines, recommendations and/or opinions released from time to time by the European Banking Authority (or any successor or replacement thereof) or the Relevant Regulator, as the case may be.

"CRR" means the Regulation (2013/575) of the European Parliament and of the Council on prudential requirements for credit institutions and investment firms dated 26 June 2013 and published in the Official Journal of the European Union on 27 June 2013, as amended or replaced from time to time (including, for the avoidance of doubt, the amendments to such Regulation resulting from Regulation (EU) 2019/876 of the European Parliament and of the Council as regards the leverage ratio, the net stable funding ratio, requirements for own funds and eligible liabilities, counterparty credit risk, market risk, exposures to central counterparties, exposures to collective investment undertakings, large exposures, reporting and disclosure requirements dated 20 May 2019 and published in the Official Journal of the European Union on 7 June 2019).

"Danish Act on Recovery and Resolution of certain Financial Businesses" means the Danish Act on Recovery and Resolution of certain Financial Businesses in force on the Faroe Islands pursuant to the decree no. 183 of 15 March 1989 and most recently amended by the decree no. 1032 of 29 June 2022 as amended or replaced from time to time.

"Danish Capital Markets Act" means the Danish Securities Trading etc., Act in force on the Faroe Islands pursuant to the decree no. 986 of 11 August 2010 and no. 993 of 14 October 2011 as amended or replaced from time to time and (if and when made effective on the Faroe Islands) the Danish Capital Markets Act (Consolidated Act no. 41 of 13 January 2023, as amended or replaced from time to time).

"Danish Statutory Loss Absorption Powers" means any write-down, conversion, transfer, modification, suspension or similar or related power existing from time to time under, and exercised in compliance with, any laws, regulations, rules or requirements in effect in Denmark, relating to (i) the transposition of the BRRD (or, as the case may be, any provision of Danish or Faroese law transposing or implementing the BRRD (or, as the case may be, any provision of Danish or Faroese law transposing or implementing the BRRD) as amended or replaced from time to time and (ii) the instruments, rules and standards created thereunder, pursuant to which any obligation of the Issuer (or any affiliate of the Issuer) can be reduced, cancelled, modified or converted into other Securities or other obligations of the Issuer or any other person (or suspended for a temporary period).

"DKK" means the lawful currency of the Kingdom of Denmark.

"Enforcement Event" has the meaning as ascribed to it in Clause 13 (Enforcement Events).

"Euronext Securities Copenhagen" means VP Securities A/S (Business Registration CVR 2159 9336), Nicolai Eigtveds Gade 8, DK-1402 Copenhagen K, Denmark.

"First Call Date" means the Interest Payment Date falling on or immediately after the fourth anniversary of the Issue Date.

"Group" means the Issuer together with its subsidiaries and other entities, if any, that are consolidated in the Issuer's calculation of the MREL Requirement on a consolidated level in accordance with the CRD/CRR requirements.

"Independent Adviser" means an independent financial institution of repute in the debt capital markets where the CIBOR Rate is commonly used or other independent financial adviser experienced in the debt capital markets where the CIBOR Rate is commonly used, in each case appointed by the Issuer at its own expense.

"Interest Determination Date" has the meaning as ascribed to it in Clause 8.1.4.

"Interest Payment Date" means each date pursuant to which interest is payable pursuant to Clause 8.1.1.

"Interest Period" means the period beginning on (and including) the Issue Date and ending on (but excluding) the first Interest Payment Date and each successive period beginning on (and including) and Interest Payment Date and ending on (but excluding) the next succeeding Interest Payment Date.

"Issue Date" has the meaning ascribed to it in Clause 2.4.

"Issuer" has the meaning ascribed to it in Clause 2.1.

"Issuing Agent" has the meaning ascribed to it in Clause 2.5.

"Margin" has the meaning ascribed to it in Clause 8.1.3(b).

"**MREL Disqualification Event**" means, in respect of the Notes, the determination by the Issuer that, as a result of:

(a) the implementation of any Applicable MREL Regulations on or after the Issue Date; or

(b) a change in any Applicable MREL Regulations becoming effective on or after the Issue Date,

all or part of the Outstanding Principal Amount of the Notes will be excluded from the "eligible liabilities" (or any equivalent or successor term) available to meet any MREL Requirement (however called or defined by then Applicable MREL Regulations) if the Issuer and/or the Group is/are then or, as the case may be, will be subject to such MREL Requirement, provided that an MREL Disqualification Event shall not occur where such exclusion is or will be caused by (i) the remaining maturity of such Notes being less than any period prescribed by any applicable eligibility criteria under the Applicable MREL Regulations, or (ii) any applicable limits on the amount of "eligible liabilities" (or any equivalent or successor term) permitted or allowed to meet any MREL Requirement(s) being exceeded. "**MREL Eligible Liabilities**" means "eligible liabilities" (Da: *nedskrivningsegnede passiver*) (or any equivalent or successor term) which are available to meet any MREL Requirement (however called or defined by then Applicable MREL Regulations) of the Issuer and / or the Group under the Applicable MREL Regulations.

"**MREL Requirement**" means the minimum requirement for own funds and eligible liabilities, in each case which is or, as the case may be, will be, applicable to the Issuer and / or the Group;

"**Modified Following Business Day Convention**" means that the relevant date shall be postponed to the first following day that is a Business Day unless that day falls in the next calendar month in which case that date will be the first preceding day that is a Business Day.

"**Non-Preferred Senior Obligations**" means any unsubordinated and unsecured liabilities of the Issuer which rank upon an insolvency of the Issuer in accordance with Section 13(3) of the Danish Act on Recovery and Resolution of certain Financial Businesses.

"**Noteholders**" means a person who is registered in Euronext Securities Copenhagen as directly registered owner or nominee holder of a Note.

"Noteholders' Meeting" means a Noteholders' meeting held pursuant to Clause 15 (Noteholders' Meeting).

"Notes" has the meaning ascribed to it in Clause 2.1.

"Outstanding Principal Amounts" means, in respect of a Note or the Notes, as the context requires, the outstanding principal amount thereof, as adjusted from time to time for any reduction of the principal amount as required by then current legislation and/or regulations applicable to the Issuer.

"Paying Agent" has the meaning ascribed to it in Clause 2.6.

"Permission Withdrawal Early Redemption Restriction" has the meaning ascribed to it in Clause 11.8.2.

"Qualifying Preferred Senior Notes" means, at any time, any securities (other than the Notes) issued or guaranteed by the Issuer that:

- (a) contain terms which at such time comply with the then current requirements for "eligible liabilities" (or any equivalent or successor term) provided for in the Applicable MREL Regulations in relation to the relevant MREL Requirement (which, for the avoidance of doubt, may result in such securities not including, or restricting for a period of time the application of, one or more of the early redemption rights which are included in the Notes);
- (b) carry the same rate of interest from time to time applying to the Notes prior to the relevant substitution or variation pursuant to Clause 11.7 (*Substitution and variation*); and
- (c) have the same currency of payment, denomination, original principal amount and Outstanding Principal Amounts as the Notes prior to the relevant substitution or variation pursuant to Clause 11.7 (Substitution and variation);
- (d) have the same Maturity Date and the same Interest Payment Dates as the Notes prior to the relevant substitution or variation pursuant to Clause 11.7 (*Substitution and variation*);
- (e) have at least the same ranking as the Notes prior to the relevant substitution or variation pursuant to Clause 11.7 (*Substitution and variation*);

- (f) shall not immediately following the relevant substitution or variation pursuant to Clause 11.7 (*Substitution and variation*) be subject to an MREL Disqualification Event and / or Tax Event;
- (g) have terms not otherwise materially less favourable to the Noteholders than the terms of the Notes, as reasonably determined by the Issuer, and provided that the Issuer shall have delivered a certificate to that effect signed by authorised signatories of the Issuer to the Issuing Agent (and copies thereof will be available at the Issuing Agent's specified office during its normal business hours) not less than five (5) Business Days prior to (x) in the case of a substitution of the Notes pursuant to Clause 11.7 (*Substitution and variation*), the issue date of the relevant securities or (y) in the case of a variation of the Notes pursuant to Clause 11.7 (*Substitution and variation*), the issue 11.7 (*Substitution and variation*), the date such variation becomes effective; and
- (h) if (A) the Notes were listed or admitted to trading on a regulated market immediately prior to the relevant substitution or variation, are listed or admitted to trading on a regulated market or (B) the Notes were listed or admitted to trading on a recognized stock exchange other than a regulated market immediately prior to the relevant substitution or variation, are listed or admitted to trading on any recognised stock exchange (including, without limitation, a regulated market), in either case as selected by the Issuer.

"**Reference Banks**" means the principal Danish office of four major banks engaged in the Copenhagen interbank market as selected by the Calculation Agent after consultation with the Issuer.

"**Relevant Date**" means the date on which a payment in respect of the Notes first becomes due, except that, if the full amount of the moneys payable has not been duly received by a Paying Agent on or before the due date, it means the date on which, the full amount of such moneys having been so received, notice to that effect is duly given to the Noteholders in accordance with Clause 22 (*Notices*).

"**Relevant Nominating Body**" means in relation to the CIBOR Rate (or such Successor CIBOR Rate or Alternative CIBOR Rate which has been determined in relation to the CIBOR Rate pursuant to the operation of Clause 8.2):

- (a) the administrator of the reference rate, or any entity under the common control as the administrator of the reference rate;
- (b) the central bank for the currency to which the reference rate relates, or any central bank or other supervisory authority which is responsible for supervising the administrator of the reference rate; or
- (c) any working group or committee sponsored by, chaired or co-chaired by or constituted at the request of (i) the central bank for the currency to which the reference rate relates, (ii) any central bank or other supervisory authority which is responsible for supervising the administrator of the reference rate, (iii) a group of the aforementioned central banks or other supervisory authorities, or (iv) the Financial Stability Board or any part thereof.

"Relevant Regulator" means, in relation to the Issuer or the Group, as the case may be, the Danish Financial Supervisory Authority (Da: *Finanstilsynet*) and any successor or replacement thereto, and/or such other authority having primary responsibility for the prudential oversight and supervision of the Issuer or the Group and / or the Relevant Resolution Authority (if applicable), in any case as determined by the Issuer.

"**Relevant Resolution Authority**" means the resolution authority with the ability to exercise any Danish Statutory Loss Absorption Powers (or any other power under the BRRD) in relation to the Issuer and / or the Group.

"**Representative Amount**" means an amount that is representative for a single transaction in the relevant market at the relevant time.

"Securities" means any securities, including shares, of the Issuer.

"Successor CIBOR Rate" means the rate that an Independent Adviser determines is a successor to or the replacement of the applicable CIBOR Rate and which is formally recommended by a Relevant Nominating Body.

"**Tax Event**" means as a result of any change in the laws, regulations or rulings of Faroese tax authorities (TAKS), or any authority in, or of, the Faroe Islands having the power to tax or in the official interpretation or administration of any such laws, regulations or rulings on or after the Issue Date, the Issuer receives an opinion of external counsel in the Faroe Islands experienced in such matters that (A) it would be required to pay Additional Amounts as provided in Clause 12 (*Taxation (Gross up)*) or (B) it will no longer be able to obtain a tax deduction for the purposes of Faroese tax for any payment of interest under the Notes (or the value of such deduction would be materially reduced), in each case provided that the Issuer satisfies the Relevant Regulator that such change in tax treatment of the Notes is material and was not reasonably foreseeable at the time of their issuance.

"Written Procedure" means a written procedure held pursuant to Clause 16 (Written Procedure).

4. Form, denomination, nominal amount, trades

- 4.1 The Notes are issued in bearer form (Da: *ihændehaver*) and issued in uncertificated and dematerialised book-entry form through Euronext Securities Copenhagen.
- 4.2 The Notes are denominated in DKK. The Notes shall be registered in Euronext Securities Copenhagen in multiples of DKK 1,000,000. All trades in Notes as well as the initial subscription for Notes shall be in a minimum amount of DKK 1,000,000 and, if more, in even multiples of DKK 1,000,000.
- 4.3 If a Noteholder holds a nominal amount in a custody account of less than DKK 1,000,000, such Notes may not be traded unless such Noteholder purchases or transfers additional Notes in the custody account so that the requirements as to tradeable amounts of DKK 1,000,000 are satisfied.
- 4.4 The Outstanding Principal Amounts may be adjusted as required by then current mandatory legislation and/or regulations applicable to the Issuer. Any such adjustment to the Outstanding Principal Amounts will not have any effect on the denominations of the Notes.
- 4.4.1 The ISIN code of the Notes is: DK0030529664.

5. Transferability and title

5.1 The Notes are freely transferable, but the Noteholders may be subject to purchase or transfer restrictions with regard to the Notes under Clauses 4.2 and 4.3 or under laws to which a Noteholder may be subject. Each Noteholder must ensure compliance with such restrictions at its own cost and expense.

- 5.2 Legal title to the Notes will pass by electronic registration in the book entry system and register maintained by Euronext Securities Copenhagen in accordance with the Danish Capital Markets Act, Executive Orders issued pursuant thereto and the rules and procedures of Euronext Securities Copenhagen from time to time. Each Noteholder shall (except as otherwise required by law) be treated as holder of the Notes for all purposes and no person shall be liable for so treating such Noteholder.
- 5.3 The Issuer shall, to the extent permitted under applicable laws, regulations, and the rules and procedures of Euronext Securities Copenhagen from time to time, have access on demand to static data and ownership of the Noteholders registered in the securities register.

6. Noteholders' rights

- 6.1 If a beneficial owner of a Note not being registered as a Noteholder wishes to exercise any rights under the Notes (including, but not limited to participating in Noteholders' Meeting or a Written Procedure), it must obtain proof of ownership of the Notes, acceptable to the chairman of the Noteholders' Meeting (in case of a Noteholders' Meeting) or the Issuer (in case of a Written Procedure).
- 6.2 A Noteholder (whether registered as such or proven to the satisfaction of the chairman of the Noteholders' Meeting or the Issuer, as applicable, to be the beneficial owner of the Note as set out in Clause 6.1) may issue one or more powers of attorney to third parties to represent it in relation to some or all of the Notes held or beneficially owned by such Noteholder. The chairman of the Noteholders' Meeting or the Issuer, as applicable, shall only have to examine the face of a power of attorney or similar evidence of authorisation that has been provided to it pursuant to this Clause 6 and may assume that it is in full force and effect, unless otherwise is apparent from its face or the chairman of the Noteholders' Meeting or the Issuer, as applicable, has actual knowledge to the contrary.

7. Status of the Notes

- 7.1 The Notes on issue constitute Preferred Senior Obligations of the Issuer.
- 7.2 The Notes constitute direct unsubordinated and unsecured debt obligations of the Issuer, and shall at all times rank:
 - (a) *pari passu* without any preference among themselves;
 - (b) at least *pari passu* with all other outstanding senior, unsecured and unsubordinated obligations of the Issuer (save for obligations which may be preferred by law, including obligations benefitting from a preferred ranking to the Notes), present and future, without any preference by reason of priority of date of creation, currency of payment or otherwise as regards the right to receive periodic payments on a liquidation or bankruptcy of the Issuer and the right to receive repayment of capital on a liquidation or bankruptcy of the Issuer; and
 - (c) senior to any Non-Preferred Senior Obligations of the Issuer as regards the right to receive periodic payments on a liquidation or bankruptcy of the Issuer and the right to receive repayment of capital on a liquidation or bankruptcy of the Issuer.

- 7.3 No Noteholder, who shall in the event of the liquidation or bankruptcy of the Issuer be indebted to the Issuer, shall be entitled to exercise any right of set-off, netting or counterclaim against moneys owed by the Issuer in respect of the Notes held by such Noteholder.
- 7.4 The Issuer reserves the right in the future to issue other notes or capital instruments, with identical or other ranking than the Notes.

8. Interest

- 8.1 Interest rate
- 8.1.1 The Notes bear interest on their Outstanding Principal Amount at a floating rate from (and including) the Issue Date.
- 8.1.2 Interest shall be payable quarterly in arrears on 22 February, 22 May, 22 August and 22 November, commencing on 22 February 2024 adjusted in accordance with the Modified Following Business Day Convention in accordance with this Clause 8.1 and Clause 10 (*Payments*).
- 8.1.3 The rate of interest payable from time to time in respect of the Notes will be determined as the sum of:
 - (a) the Copenhagen interbank offered rate (CIBOR) per annum, administered by the Danish Financial Benchmark Facility ("DFBF") (or any other person which takes over administration of that rate) for three (3) months deposits in DKK as displayed on the DFBF webpage https://dfbf.dk/dfbf-benchmarks/rates/ (or any replacement webpage of DFBF which displays that rate) or on the appropriate page of such other information service which publishes that rate from time to time in place of DFBF (and if the agreed page is replaced or services ceases to be available, the Calculation Agent may specify another page or service displaying the appropriate rate) as of around 11:00 am (Copenhagen time) (the "CIBOR Rate"); and
 - (b) 2.25 per cent per annum (the "Margin").
- 8.1.4 The CIBOR Rate is fixed by the Calculation Agent two (2) Business Days before the start of an Interest Period (each an "Interest Determination Date").
- 8.1.5 The Calculation Agent will cause the rate of interest and each interest amount for each Interest Period and the relevant Interest Payment Date to be notified to the Issuer, the Paying Agent, Euronext Securities Copenhagen and any stock exchange or other relevant authority on which the Notes are for the time being listed or by which they have been admitted to trading (by no later than the first day of each Interest Period) and notice thereof to be published in accordance with Clause 22 (*Notices*) as soon as possible after their determination but in no event later than the fourth Business Day thereafter. Each interest amount and Interest Payment Date so notified may subsequently be amended (or appropriate alternative arrangements made by way of adjustment) without prior notice in the event of an extension or shortening of the Interest Period. Any such amendment will be promptly notified to each stock exchange on which the Notes are for the time being listed and to the Noteholders in accordance with Clause 22 (*Notices*).
- 8.1.6 Interest accrues from day to day and is calculated on the basis of the actual number of days in the interest period divided by 360 (Actual/360, adjusted).

8.2 Unavailability of CIBOR Rate

- 8.2.1.1 If the CIBOR Rate is unavailable, the Calculation Agent will request each of the Reference Banks to provide the Calculation Agent with:
 - (a) (other than where paragraph (b) below applies) the rate at which the relevant Reference Bank is willing to lend amounts in DKK for three months without collateral to other banks active on the Danish money market; or
 - (b) if different, the rate (if any and applied to the relevant Reference Bank and the relevant period) which contributors to the CIBOR Rate are asked to submit to the relevant administrator, in each case at approximately 11.00 a.m. (Copenhagen time) on the second Business Day prior to the start of each Interest Period.
- 8.2.1.2 If at least two of the Reference Banks provide such rates, the rate of interest shall be the arithmetic mean (rounded if necessary to the fifth decimal place, with 0.000005 being rounded upwards) as established by the Calculation Agent of such rates, plus the Margin. If fewer than two rates are provided as requested, the Interest Rate for that Interest Period will be the arithmetic mean of the rates quoted by major banks in Denmark selected by the Calculation Agent, at approximately 11.00a.m. (Copenhagen time) on the first day of such Interest Period for loans in DKK to leading Danish banks for a period of three months commencing on the first day of such Interest Period and for a Representative Amount, plus the Margin.
- 8.2.1.3 If the Interest Rate cannot be determined in accordance with the above provisions, the Interest Rate shall be the Interest Rate applicable to the preceding Interest Period, all as determined by the Calculation Agent.
- 8.3 Determination of Interest Rate binding
- 8.3.1 All notifications, opinions, determinations, certificates, calculations, quotations and decisions given, expressed, made or obtained for the purposes of this Clause 8 by the Calculation Agent shall (in the absence of manifest error) be binding on the Calculation Agent, the Issuer, the Paying Agent and all Noteholders and (in the absence of wilful default and gross negligence) no liability to the Noteholders, the Issuer and the Paying Agent shall attach to the Calculation Agent in connection with the exercise or non-exercise by it of any of its powers, duties and discretions.

9. CIBOR Rate replacement

9.1 General

Any determination to be made by or any changes to the Terms and Conditions to be specified by the Independent Adviser in accordance with the provisions of this Clause 9 shall at all times be made by such Independent Adviser acting in good faith.

- 9.2 CIBOR Rate determination
- 9.2.1 If (i) the CIBOR Rate, from time to time, has ceased to be published on the relevant website for at least five (5) consecutive Business Days or the CIBOR Rate has ceased to exist as a result of the CIBOR Rate ceasing to be calculated or administered, (ii) the administrator of the CIBOR Rate has made a public statement or publication of information announcing that within six (6) months it will

cease to provide the CIBOR Rate permanently or indefinitely, or (iii) a Relevant Nominating Body has made a public statement or publication of information recommending the usage of a Successor CIBOR Rate for the applicable CIBOR Rate (which better reflects the relevant market interest rates), a "CIBOR Rate Event" has occurred and the following shall apply:

- (a) the Issuer shall use reasonable endeavours to appoint, as soon as reasonably practicable, an Independent Adviser to determine, no later than five (5) Business Days prior to the relevant Interest Determination Date in relation to the next succeeding Interest Period (the "CIBOR Rate Determination Date"), a Successor CIBOR Rate or, alternatively, if there is no Successor CIBOR Rate, an Alternative CIBOR Rate for purposes of determining the applicable CIBOR Rate for the next succeeding Interest Period;
- (b) if the Issuer is unable to appoint an Independent Adviser, or the Independent Adviser appointed by it fails to determine a Successor CIBOR Rate or an Alternative CIBOR Rate prior to a CIBOR Rate Determination Date, the CIBOR Rate applicable to the next succeeding Interest Period shall be equal to the CIBOR Rate last determined for the preceding Interest Period; and
- (c) if a Successor CIBOR Rate or an Alternative CIBOR Rate is determined in accordance with paragraph 9.2.1 in this Clause 9.2.1, such Successor CIBOR Rate or Alternative CIBOR Rate shall be the CIBOR Rate for each of the future Interest Periods (subject to the subsequent operation of, and to adjustment as provided for in this Clause 9).
- 9.2.2 If the Independent Adviser (in consultation with the Issuer), determines that an Adjustment Spread is required to be applied to the Successor CIBOR Rate or the Alternative CIBOR Rate and that such Adjustment Spread is determined by the Independent Adviser, such Adjustment Spread shall be applied.
- 9.3 Variation upon CIBOR Rate replacement
- 9.3.1 If the Independent Adviser determines a Successor CIBOR Rate or an Alternative CIBOR Rate in accordance with this Clause 9.2, the Independent Adviser may (following consultation and subject to the final approval by the Issuer) specify changes to these Terms and Conditions in order to follow market practice in relation to the relevant Successor CIBOR Rate or Alternative CIBOR Rate.
- 9.3.2 The Issuer shall, at its own expense, but subject to receipt by the Noteholders of the certificate referred to in Clause 9.3.3, without the requirement for any consent or approval of the Noteholders, effect such amendments to these Terms and Conditions as may be required by the Issuer in order to give effect to this Clause 9, such amendments referred to as "**CIBOR Rate Amendments**".
- 9.3.3 The Issuer shall promptly, following the determination of any Successor CIBOR Rate or Alternative CIBOR Rate and any CIBOR Rate Amendments, give notice thereof to the Calculation Agent and the Noteholders in accordance with Clause 22 (*Notices*). No later than giving the Noteholders such notice, the Issuer shall deliver to the Noteholders a certificate signed by two (2) authorised signatories of the Issuer:
 - (a) confirming (i) that a CIBOR Rate Event has occurred, (ii) the relevant Successor CIBOR Rate or Alternative CIBOR Rate, and (iii) any CIBOR Rate Amendments, in each case as determined in accordance with the provisions of this Clause 9; and

- (b) as applicable, certifying that the CIBOR Rate Amendments are necessary to ensure the proper operation of such Successor CIBOR Rate or Alternative CIBOR Rate.
- 9.3.4 The Successor CIBOR Rate or Alternative CIBOR Rate and any CIBOR Rate Amendments specified in such certificate will (in the absence of manifest error or bad faith in the determination of the Successor CIBOR Rate or Alternative CIBOR Rate and any CIBOR Rate Amendments and without prejudice to the Calculation Agent's ability to rely on such certificate as aforesaid) be binding on the Issuer, the Calculation Agent and the Noteholders.
- 9.3.5 An Independent Adviser appointed pursuant to this Clause 9 shall act in good faith and (in the absence of bad faith or fraud) shall have no liability whatsoever to the Paying Agent, Calculation Agent or the Noteholders for any advice given to the Issuer in connection with any determination made by the Issuer pursuant to this Clause 9.
- 9.3.6 Notwithstanding any other provision of this Clause 9, no Successor CIBOR Rate or Alternative CIBOR Rate (as applicable) will be adopted, and no other amendments to the terms of the Notes will be made pursuant to this Clause 8, if and to the extent that, in the determination of the Issuer, the same could reasonably be expected to prejudice the qualification of the Notes as MREL Eligible Liabilities of the Issuer and/or the Group.

10. Payments

10.1 Payments of principal and interest

Payments of principal, interest and any other amounts in respect of the Notes shall be made to the Noteholders shown in the relevant records of Euronext Securities Copenhagen in accordance with and subject to the rules and regulations from time to time governing Euronext Securities Copenhagen.

10.2 Payments subject to fiscal laws

All payments in respect of the Notes are subject in all cases to any applicable fiscal or other laws and regulations in the place of payment, but without prejudice to the provisions of Clause 12 (*Taxation (Gross up)*). No commissions or expenses shall be charged to the Noteholders in respect of such payments.

10.3 Payments on Business Days

If the due date for payment of any amount in respect of any Note is not a Business Day, the payment shall be postponed to the following Business Day, and the Noteholders shall not be entitled to any further interest or other payment in respect of such delay.

11. Redemption and purchase

11.1 Redemption at maturity

Unless previously redeemed or purchased and cancelled, the Notes will be redeemed at their Outstanding Principal Amount on the Maturity Date in accordance with Clause 10 (*Payments*)

11.2 Early redemption upon the occurrence of a Tax Event

Subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*), upon the occurrence of a Tax Event, the Issuer may, at its option at any time, and having given no less than thirty (30) and no more than sixty (60) days' notice to the Paying Agent and the Noteholders in accordance with Clause 22 (*Notices*) (which notice shall, subject to the Permission Withdrawal Early Redemption Restriction, be irrevocable), redeem all (but not some only) of the outstanding Notes at their Outstanding Principal Amounts, together with accrued and unpaid interest thereon.

11.3 Early redemption upon the occurrence of an MREL Disqualification Event

Subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*), upon the occurrence of an MREL Disqualification Event, the Issuer may, at its option at any time, and having given no less than thirty (30) and no more than sixty (60) days' notice to the Paying Agent and the Noteholders in accordance with Clause 22 (*Notices*) (which notice shall, subject to the Permission Withdrawal Early Redemption Restriction, be irrevocable), redeem all (but not some only) of the outstanding Notes at their Outstanding Principal Amounts, together with accrued and unpaid interest thereon.

11.4 Redemption at the option of the Issuer

Subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*), the Issuer may, at its option and having given no less than fifteen (15) and no more than thirty (30) days' notice to the Paying Agent and the Noteholders in accordance with Clause 22 (*Notices*) (which notice shall, subject to the Permission Withdrawal Early Redemption Restriction, be irrevocable), redeem all (but not some only) of the outstanding Notes on the relevant Call Date, and at any Interest Payment Date thereafter, at their Outstanding Principal Amounts, together with accrued and unpaid interest thereon.

11.5 Purchase of Notes by the Issuer and any other member of the Group

The Issuer and any other member of the Group may (subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*)) purchase Notes in the open market or otherwise at any price.

11.6 Cancellation

All Notes purchased by or on behalf of the Issuer may (but subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*)) be cancelled by the Issuer when the Issuer holds the title to them. The Notes are cancelled in the records of Euronext Securities Copenhagen so that the cancelled Notes cannot be reissued or resold, and subsequently the Issuer has no obligations pertaining to the cancelled Notes. The outstanding amount of Notes will be updated in the records of Euronext Securities Copenhagen.

11.7 Substitution and variation

11.7.1 Subject to having given no less than thirty (30) nor more than sixty (60) days' notice (which notice shall be irrevocable) to the Noteholders (in accordance with Clause 22 (Notices)) and the Issuing Agent:

- (a) if a MREL Disqualification Event and/or a Tax Event has/have occurred and is/are continuing; or
- (b) in order to align the Terms and Conditions to best practices published by the European Banking Authority (or any successor or replacement thereof) resulting from its monitoring activities pursuant to Article 80 of the CRR,

the Issuer may, at its option, but subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*), substitute all (but not some only) of the Notes or vary the terms of all (but not some only) of the Notes, without any requirement for the consent or approval of the Noteholders, so that they become or remain Qualifying Preferred Senior Notes.

- 11.7.2 Any such notice shall specify the relevant details of the manner in which such substitution or variation shall take effect and where the Noteholders can inspect or obtain copies of the new terms and conditions of the Qualifying Preferred Senior Notes. Such substitution or variation will be effected without any cost or charge to the Noteholders.
- 11.8 Conditions to redemption etc. prior to the Maturity Date
- 11.8.1 The Notes may only be redeemed, purchased, cancelled, substituted, varied or modified (as applicable) pursuant to Clauses 11.2 (*Early redemption upon the occurrence of a Tax Event*), 11.3 (*Early redemption upon the occurrence of an MREL Disqualification Event*), 11.4 (*Redemption at the option of the Issuer*), 11.5 (*Purchase of Notes by the Issuer and any other member of the Group*), 11.6 (Cancellation), 11.7 (Substitution and variation), 14.1 (*Powers of Noteholders' Meeting and a Written Procedure*) and 18.1(b), as the case may be, if:
 - (a) in the case of any substitution, variation or modification, the Issuer has notified the Relevant Regulator of, and the Relevant Regulator has not objected to such substitution, variation or modification (as applicable) in accordance with the CRD/CRR and BRRD requirements;
 - (b) in the case of any such substitution, redemption, purchase or cancellation, the Issuer has notified the Relevant Regulator of, and the Relevant Regulator has granted its permission to, such redemption, purchase or cancellation (as applicable) and, if so given by the Relevant Regulator, such permission has not been withdrawn by the Relevant Regulator prior to the date fixed for substitution, redemption, purchase or cancellation (as applicable)in accordance with the CRD/CRR and BRRD requirements (which, as at the Issue Date, are set out in Articles 77 and 78a of the CRR); and
 - (c) in the case of a redemption of the Notes as a result of a Tax Event or an MREL Disqualification Event, the Issuer has delivered a certificate signed by two of its authorised signatories to the Issuing Agent (and copies thereof will be available at the Issuing Agent's specified office during its normal business hours) not less than five (5) Business Days prior to the date set for redemption that such Tax Event or MREL Disqualification Event has occurred or will occur no more than ninety (90) days following the date fixed for redemption, as the case may be.
- 11.8.2 If, after a notice of redemption has been given pursuant to Clauses 11.2 (*Early redemption upon the occurrence of a Tax Event*), 11.3 (*Early redemption upon the occurrence of an MREL Disqualification Event*), 11.4 (*Redemption at the option of the Issuer*), (as applicable), the Relevant Regulator withdraws its permission to the relevant redemption before the relevant redemption date, such notice of redemption shall automatically be revoked and the relevant redemption shall not be made until a new redemption notice is given and all conditions for redemption as described in this Clause 11.8.2

have been fulfilled. The redemption restriction described in this Clause 11.8.2 is referred to as the **"Permission Withdrawal Early Redemption Restriction**".

- 11.8.2.1 If the Issuer has elected to substitute or vary the Notes pursuant to Clause 11.7 (*Substitution and variation*) but prior to the relevant substitution or variation, as the case may be, the Relevant Regulator withdraws its no objection to the relevant substitution or variation (as applicable) the relevant notice shall be automatically rescinded and shall be of no force and effect.
- 11.8.3 Any refusal by the Relevant Regulator to grant its permission to any such redemption, purchase or cancellation (as applicable) pursuant to Clause 11.8.1(b) or any objection by the Relevant Regulator to any such variation, substitution or modification (as applicable) pursuant to Clause 11.8.1(a) (or, as the case may be, any withdrawal by the Relevant Regulator of any such permission or no objection (as applicable)) will not constitute an event of default or an Enforcement Event under the Notes.

12. Taxation (Gross up)

- 12.1 All payments in respect of the Notes by or on behalf of the Issuer shall be made without withholding or deduction for, or on account of, any present or future taxes, duties, assessments or governmental charges of whatever nature (together, "**Taxes**") imposed or levied by or on behalf of the Faroese tax authorities (TAKS), or any authority in, or of, the Faroe Islands having the power to tax, unless the withholding or deduction of the Taxes is required by law. In that event, in the case of a payment of interest only, the Issuer will pay such additional amounts ("**Additional Amounts**") as may be necessary in order that the net amounts received by the Noteholders after such withholding or deduction shall equal the amounts which would have been receivable in respect of the Notes in the absence of the withholding or deduction; except that no Additional Amounts shall be payable in relation to any payment in respect of any Note:
 - (a) to, or to a third party on behalf of, a Noteholder who is liable to the Taxes in respect of the Note by reason of him having some connection with the Faroe Islands other than the mere holding of the Note or receipt of interest in respect thereof; or
 - (b) presented for payment more than thirty (30) days after the Relevant Date except to the extent that a Note would have been entitled to Additional Amounts on claiming payment on or before the expiry of such period of thirty (30) days.
- 12.2 Notwithstanding any other provision of the Terms and Conditions, in no event will the Issuer be required to pay any additional amounts in respect of the Notes for, or on account of, any withholding or deduction required pursuant to an agreement described in section 1471(b) of the U.S. Internal Revenue Code of 1986 (the "**Code**") or otherwise imposed pursuant to sections 1471 through 1474 of the Code, any regulations or agreements thereunder, or any official interpretations thereunder.

13. Enforcement Events

13.1 No events of default

There are no events of default in respect of the Notes. Noteholders shall not be entitled at any time to file for bankruptcy or liquidation of the Issuer.

13.2 Enforcement Events

If an order is made or an effective resolution is passed for the bankruptcy or liquidation of the Issuer (an "**Enforcement Event**"), any Noteholder may prove or claim in such proceedings in respect of such Note, such claim being for payment of the Outstanding Principal Amount of such Note at the time of commencement of such bankruptcy or liquidation of the Issuer together with any interest accrued and unpaid on such Note (to the extent that the same is not cancelled in accordance with the terms of the Notes) from (and including) the Interest Payment Date immediately preceding the occurrence of such Enforcement Event and any other amounts payable on such Note (including any damages payable in respect thereof). Such claim shall rank as provided in Clause 7 (*Status of the Notes*).

13.3 Enforcement of obligations

Subject to Clause 13.1 (*No events of default*) and without prejudice to Clause 13.2 (*Enforcement Events*), any Noteholder may, at its discretion and without further notice, institute such proceedings against the Issuer as it may think fit to enforce any obligation, condition or provision binding on the Issuer under the Notes, provided that the Issuer shall not by virtue of the institution of any proceedings be obliged to pay any sum or sums sooner than the same would otherwise have been payable by it.

14. Decisions by Noteholders

14.1 Powers of Noteholders' Meetings and a Written Procedure

A Noteholders' Meeting or a Written Procedure shall, subject to the Terms and Conditions (including Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*)), have power to:

- (a) sanction any proposal by the Issuer for any modification, abrogation, variation or compromise of, or arrangement in respect of, the rights of the Noteholders against the Issuer, whether or not those rights arise under the Notes;
- (b) sanction the exchange or substitution for the Notes of, or the conversion of the Notes into, Securities, Notes or other obligations of the Issuer or any other entity;
- (c) assent to any modification of the Notes or the Terms and Conditions proposed by the Issuer;
- (d) to authorise anyone to concur in and do anything necessary to carry out and give effect to a resolution taken at a Noteholders' Meeting or a Written Procedure;
- (e) appoint and elect a representative on behalf of the Noteholders pursuant to any provision of the Danish Capital Markets Act in force on the Faroe Islands;
- (f) appoint any persons (whether Noteholders or not) as a committee or committees to represent the Noteholders' interests and to confer on them any powers or discretions which the Noteholders could themselves exercise at a Noteholders' Meeting or a Written Procedure; and
- (g) approve the substitution of any entity for the Issuer (or any previous substitute) as principal debtor under the Notes or the Terms and Conditions.
- 14.1.2 Decisions to be taken by the Noteholders may be dealt with, at the option of the Issuer, at a Noteholders' Meeting or by way of a Written Procedure.

- 14.1.3 A Noteholders' Meeting will be held in accordance with the procedure pursuant to Clause 15 (*Noteholders' Meeting*).
- 14.1.4 A Written Procedure will be held in accordance with the procedure pursuant to Clause 16 (*Written Procedure*).
- 14.2 Voting rights
- 14.2.1 Each Noteholder holds one vote for each Note. The Issuer has no voting rights in respect of Notes held by the Issuer or any of its subsidiaries.
- 14.2.2 Only a person who is, or who has been provided with a power of attorney from a person who is, able to document its holdings of Notes by
 - (a) presenting a custody account statement from Euronext Securities Copenhagen or an authorised institution that is not more than three Business Days old (where the three Business Days shall be counted from the date of the submission of the vote or power of attorney authorising a person to vote); or
 - (b) provide other proof of holding which, in the case of a Noteholders' Meeting is satisfactory to the chairman of the Noteholders' Meeting or in the case of a Written Procedure is satisfactory to the Issuer, may exercise voting rights as a Noteholder at such Noteholders' Meeting or in such Written Procedure.
 - (c) For the purposes of this Clause 14.2, a beneficial owner of a Note that has a Note registered in the name of a nominee will, in accordance with Clause 6 (*Noteholders' rights*), be deemed to be the owner of the Note rather than the nominee. No vote may be exercised at Noteholders' Meeting or in a Written Procedure by any nominee if the beneficial owner of the Note has presented relevant evidence to the chairman of the Noteholders' Meeting (in case of a Noteholders' Meeting) or the Issuer (in case of a Written Procedure) pursuant to Clause 6 (*Noteholders' rights*) stating that it is the beneficial owner of the Notes voted for. If such owner of the Notes has voted directly for any of its nominee registered Notes, the owner of the Notes votes shall take precedence over votes submitted by the nominee for the same Notes.
- 14.3 Percentage of Noteholders required to consent
- 14.3.1 The following matters shall require the consent of Noteholders representing at least 66 2/3 per cent of the Outstanding Principal Amounts for which Noteholders are voting at a Noteholders' Meeting or for which Noteholders reply in a Written Procedure in accordance with the instructions given pursuant to Clause 15 (*Written Procedure*):
 - (a) a change to the terms of any provision of Clause 7 (Status of the Notes);
 - (b) a reduction of the amount payable upon the redemption or repurchase of any Note pursuant to Clause 11 (*Redemption and purchase*) other than as permitted or required by the Terms and Conditions;
 - (c) a change to the interest rate or the nominal amount of the Notes (other than as permitted or required by the Terms and Conditions);
 - (d) a change to the terms dealing with the requirements for Noteholders' consent set out in this Clause 14.3.1, Clause 14.4.1 or Clause 17.3;

- (e) a change of Issuer, an extension of the tenor of the Notes or any delay of the due date for payment of any principal or interest on the Notes;
- (f) a mandatory exchange of the Notes for other Securities; and
- (g) early redemption of the Notes, other than upon an acceleration of the Notes pursuant to Clause 13 (*Enforcement Events*), or as otherwise permitted or required by the Terms and Conditions.
- 14.3.2 Any matter not covered by Clause 14.3.1 above shall require the consent of Noteholders representing more than 50 per cent of the Outstanding Principal Amount for which Noteholders are voting at a Noteholders' Meeting or for which Noteholders reply in a Written Procedure.
- 14.4 Quorum
- 14.4.1 A quorum at a Noteholders' Meeting or in respect of a Written Procedure only exists if a Noteholder or Noteholders representing at least 50 per cent of the Outstanding Principal Amount in case of a matter pursuant to Clause 14.3.1, and otherwise 20 per cent of the Outstanding Principal Amount:
 - (a) attend the meeting in person (or appear through duly authorised representatives), in the case of a Noteholders' Meeting; or
 - (b) reply to the request, in the case of a Written Procedure.
- 14.4.2 Notes held by the Issuer or any of its subsidiaries shall not be taken into account when determining whether the required quorum has been met according to Clause 14.4.1 or Clause 17 (*Repeated Noteholders' Meeting or Written Procedure*).
- 14.4.3 No resolution may be passed if it is clear that that resolution is likely to give certain Noteholders or others an undue advantage over other Noteholders.
- 14.5 Issuer's, Paying Agent's, Issuing Agent's or Calculation Agent's consent required

Any decision which extends or increases the obligations of the Issuer, the Paying Agent, the Issuing Agent or the Calculation Agent or limits, reduces or extinguishes the rights or benefits of the Issuer, the Paying Agent, the Issuing Agent or the Calculation Agent under the Terms and Conditions shall be subject to the Issuer's, the Paying Agent's, the Issuing Agent's or the Calculation Agent's consent, as the case may be.

- 14.6 Decisions binding on all Noteholders and information to Noteholders
- 14.6.1 A matter decided at a duly convened and held Noteholders' Meeting or by way of Written Procedure is binding on all Noteholders, irrespective of them being present or represented at the Noteholders' Meeting or responding in the Written Procedure. The Noteholders that have not adopted or voted for a decision shall not be liable for any damages that this may cause other Noteholders.
- 14.6.2 Information about decisions taken at a Noteholders' Meeting or by way of a Written Procedure shall promptly be notified to the Noteholders, provided that a failure to do so shall not invalidate any decision made or voting result achieved. The minutes from the relevant Noteholders' Meeting or Written Procedure shall at the request of a Noteholder be sent to it by the Issuer.

14.7 Minutes of a Noteholders' Meeting or Written Procedure

Minutes shall be made of all resolutions and proceedings at every Noteholders' Meeting or Written Procedure and, if purporting to be signed by the chairman of that meeting or of the next succeeding meeting, shall be conclusive evidence of the matters in them. Until the contrary is proved, every meeting for which minutes have been so made and signed shall be deemed to have been duly convened and held and all resolutions passed or proceedings transacted at it to have been duly passed and transacted.

15. Noteholders' Meeting

- 15.1 Attendance at a Noteholders' Meeting:
- 15.1.1 At the Noteholders' Meeting, each Noteholder must document its holdings of Notes by presenting a custody account statement from Euronext Securities Copenhagen or an authorised account institution evidencing that such Noteholder was registered as a Noteholder on the Business Day(s) specified in the notice to convene Noteholders' Meeting pursuant to Clause **15.4.1** or by providing other proof of holding satisfactory to the chairman of the Noteholders' Meeting. The following may attend and speak at a Noteholders' Meeting:
 - (a) Noteholders and proxies;
 - (b) any beneficial owners of the Notes having presented relevant evidence to the chairman of the Noteholders' Meeting pursuant to Clause 6 (*Noteholders' rights*);
 - (c) any representative of the Noteholders appointed pursuant to the Danish Capital Markets Act;
 - (d) the chairman; and
 - (e) the Issuer, the Issuing Agent, the Paying Agent, the Calculation Agent and their respective financial and legal advisers.
- 15.1.2 No one else may attend or speak.
- 15.2 Chairman of the Noteholders' Meeting
- 15.2.1 The chairman of the Noteholders' Meeting shall be such person as the Issuer nominates or, if no nomination is made, the person elected by the Noteholders present at such meeting.
- 15.2.2 The Issuer shall upon request provide the chairman of the Noteholders' Meeting with the information available in the securities register kept by Euronext Securities Copenhagen in respect of the Notes in order to convene and hold the Noteholders' Meeting.
- 15.3 Convening a Noteholders' Meeting
- 15.3.1 The Issuer may at any time, and shall, if so requested by one or more Noteholders representing at least 10 per cent of the Outstanding Principal Amounts of the Notes convene a Noteholders' Meeting or initiate a Written Procedure. The Issuer may refrain from convening a Noteholders' Meeting or instigating a Written Procedure if (i) the suggested decision must be approved by any person in addition to the Noteholders and such person has informed the Issuer that an approval will not be given, or (ii) the suggested decision is not in accordance with applicable laws.

- 15.3.2 The Issuer shall call the Noteholders by notice to each Noteholders' Meeting no later than 14 days after having received request to convene such Noteholders' Meeting from the Noteholders containing the subject of such meeting. If the Issuer does not call the Noteholders' Meeting within the deadline, the Noteholders shall be entitled to call the Noteholders' Meeting. The notice to convene a Noteholders' Meeting shall be sent to all Noteholders registered in Euronext Securities Copenhagen at the time the notice to convene a Noteholders' Meeting is sent from Euronext Securities Copenhagen.
- 15.4 Notice to convene a Noteholders' Meeting
- 15.4.1 The notice pursuant to Clause 15.3 (*Convening a Noteholders' Meeting*) shall include the following:
 - (a) time for the Noteholders' Meeting, which must be at least 10 days but not more than 30 days after the notice to the Noteholders;
 - (b) place for the Noteholders' Meeting;
 - (c) a specification of the Business Day(s) on which a person must be registered as a Noteholder in order to be entitled to exercise voting rights;
 - (d) agenda for the meeting (including each request for a decision by the Noteholders); and
 - (e) a form of power of attorney.
- 15.4.2 Only matters that have been included in the notice may be resolved upon at the Noteholders' Meeting.
- 15.4.3 Should prior notification by the Noteholders be required in order to attend the Noteholders' Meeting, such requirement shall be included in the notice.
- 15.5 Venue for Noteholders' Meetings

All Noteholders' Meetings shall be held in the Aarhus area or the Copenhagen area as determined by the Issuer or the person convening the Noteholders' Meeting. The Issuer shall pay expenses associated with the meeting other than travel and other expenses incurred by the Noteholders which shall be borne by each individual Noteholder.

16. Written Procedure

- 16.1 Instigating a Written Procedure
- 16.1.1 The Issuer may instigate a Written Procedure at any time by sending a notice to each Noteholder registered in Euronext Securities Copenhagen at the time the notice to instigating a Written Procedure is sent from Euronext Securities Copenhagen.
- 16.1.2 A notice pursuant to Clause 16.1.1 shall include the following:
 - (a) each request for a decision by the Noteholders;
 - (b) a description of the reasons for each request;
 - (c) a specification of the Business Day(s) on which a person must be registered as a Noteholder in order to be entitled to exercise voting rights;

- (d) instructions and directions on replying to the request (including a form for such reply containing an option to vote yes or no for each request) as well as a form of power of attorney; and
- (e) the stipulated time period within which the Noteholder must reply to the request (such time period to last at least 5 Business Days from the notice sent pursuant to Clause 16.1.1).
- 16.1.3 Instructions for the voting shall be included in the notice.
- 16.2 Decisions

When the requisite majority consents in accordance with Clause 14.3 (*Percentage of Noteholders required to consent*) have been received in a Written Procedure, the relevant decision shall be deemed to be adopted even if the time period for replies in the Written Procedure has not yet expired.

17. Repeated Noteholders' Meeting or Written Procedure

- 17.1 Even if the necessary quorum set out in Clause 14.4.1 is not achieved, the Noteholders' Meeting or Written Procedure, as applicable, shall be held and voting completed for the purpose of recording the voting results in the minutes of the Noteholders' Meeting or Written Procedure, as applicable. The Issuer or the person who convened the initial Noteholders' Meeting or Written Procedure, as applicable, may, within ten Business Days of that Noteholders' Meeting or Written Procedure, as applicable, convene a repeated Noteholders' Meeting or Written Procedure, with the same agenda as the first Noteholders' Meeting or Written Procedure, as applicable.
- 17.2 The provisions and procedures regarding a Noteholders' Meetings and a Written Procedure, as set out, as applicable, in Clause 14 (*Decisions by Noteholders*), Clause 15 (*Noteholders' Meeting*) and Clause 16 (*Written Procedure*) shall apply *mutatis mutandis* to a repeated Noteholders' Meeting or Written Procedure, with the exception of the quorum requirements set out in Clause 14.4 (*Quorum*). A notice to convene for a repeated Noteholders' Meeting or Written Procedure, as applicable, shall also contain the voting results obtained in the initial Noteholders' Meeting or Written Procedure, as applicable,
- 17.3 The quorum at any such repeated Noteholder's Meeting or Written Procedure, as applicable, is one or more persons being or representing Noteholders whatever the principal amount of the Notes so held or represented, unless the business of such meeting includes consideration of a matter pursuant to Clause 14.3.1, in which case the quorum shall be one or more persons holding Notes or representing Noteholders holding Notes in principal amount of not less than 33 1/3 per cent of the Outstanding Principal Amount.
- 17.4 A repeated Noteholders' Meeting or Written Procedure, as applicable, may only be convened once for each initial Noteholders' Meeting or Written Procedure, as applicable. A repeated Noteholders' Meeting or Written Procedure, as applicable, may be convened pursuant to the procedures of a Written Procedure in accordance with Clause 16 (*Written Procedure*), even if the initial meeting was held pursuant to the procedures of a Noteholders' Meeting in accordance with Clause 15 (*Noteholders' Meeting*) and vice versa.

18. Modification

- 18.1 Subject to Clause 11.8 (*Conditions to redemption etc. prior to the Maturity Date*), as applicable, the Issuer may, without the consent of the Noteholders, make any modification to the Notes or these Terms and Conditions:
 - (a) to correct a manifest error; and
 - (b) which is not prejudicial to the interests of the Noteholders.
- 18.2 Subject as provided in these Terms and Conditions, no other modification may be made to the Notes or these Terms and Conditions except with consent of the Issuer and the sanction of a Noteholders' Meeting or a Written Procedure or as may be required by applicable laws or a court ruling or decision by a relevant authority.
- 18.3 Any such modification shall be binding on the Noteholders and any such modification shall be notified to the Noteholders in accordance with Clause 22 (*Notices*) as soon as practicable thereafter.

19. Further issues

The Issuer may from time to time, without the consent of the Noteholders, create and issue further Notes having the same Terms and Conditions as the Notes in all respects (or in all respects except for the first payment of interest, if any, on them and/or the issue date or the issue price thereof) so as to form a single series with the Notes.

20. Prescription limits

Claims against the Issuer for payment in respect of the Notes shall be subject to limitation under the Danish Limitation Act (Da: *lov om forældelse af fordringer*) and shall become void unless proceedings have been commenced or the limitation period has otherwise been suspended or interrupted pursuant to the rules of the Danish Limitation Act within 10 years (in the case of principal) or three years (in the case of interest) from the date when the creditor was entitled to claim payment within the meaning of section 2 of the Danish Limitation Act.

21. Replacement of agents

The Issuer reserves the right to appoint a successor Issuing Agent, Paying Agent or Calculation Agent in accordance with the rules and procedures of Euronext Securities Copenhagen from time to time, provided, however, that the Issuer shall at all times maintain an Issuing Agent and a Paying Agent which is authorized to act as an account holding institution (Da: *kontoførende institut*) with Euronext Securities Copenhagen and a Calculation Agent (which may be the Paying Agent).

22. Notices

All notices regarding the Notes to the Noteholders will be deemed to be validly given to the Noteholders if published in accordance with the procedures of Euronext Securities Copenhagen in force from time to time.

Noteholders shall be deemed for all purposes to have notice of the contents of any notice given to the Noteholders in accordance with this Clause 22.

23. Force majeure

- 23.1 Even in areas where a stricter statutory liability applies, neither the Issuer nor the Issuing Agent, the Paying Agent or the Calculation Agent shall be liable for losses due to:
 - (a) the breakdown of or lack of access to IT systems or damage to the data of these systems which can be attributed to paragraphs (b) to (d) below regardless of whether the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent (as relevant) itself or themselves or an external supplier is responsible for the operation of the systems;
 - (b) failures in the Issuer's, the Issuing Agent's, the Paying Agent's or the Calculation Agent's (as relevant) power supply or telecommunications, statutory intervention or administrative acts, natural disasters, war, insurrections, civil riots, sabotage, terror or vandalism (including computer viruses and backing);
 - (c) strike, lockout, boycott or blockade regardless of whether the conflict is directed at or initiated by the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent (as relevant) itself or themselves or its or their organisation and regardless of the reason for the conflict and whether the conflict affects all or part of the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent (as relevant); or
 - (d) other circumstances beyond the Issuer's, the Issuing Agent's, the Paying Agent's or the Calculation Agent's (as relevant) control.
- 23.2 If circumstances mentioned in Clause 23.1 occur, which make it impossible for the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent to comply with their obligations under these Terms and Conditions (to the extent they have any obligations under the Terms and Conditions), including (but not limited to) the Issuer's obligations to make payments under the Notes, these obligations will be suspended until the circumstances in question cease.
- 23.3 The Issuer's, the Issuing Agent's, the Paying Agent's or the Calculation Agent's exemption from liability pursuant to Clause 23.1 will not apply if:
 - (a) the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent (as relevant) should have anticipated the factor causing the loss when the agreement was entered into or should have avoided or overcome the reason for the loss; or
 - (b) the Issuer, the Issuing Agent, the Paying Agent or the Calculation Agent (as relevant) is liable for the factor causing the loss pursuant to applicable legislation.

24. Credit rating and listing

- 24.1 The Notes will not be assigned any credit rating by any credit rating agency
- 24.2 The Issuer shall use reasonable efforts to ensure that the Notes are admitted to trading on the regulated market of Nasdaq Copenhagen A/S within 120 days from the Issue Date or, if such admission to trading is not possible to obtain, admitted to trading on another regulated market.

- 24.3 The Issuer shall, following the admission to trading, use reasonable efforts to maintain the admission to trading as long as any Notes are outstanding, however not longer than up to and including the last day of which the admission to trading can reasonably, pursuant to the applicable regulations of the regulated market and the regulations of Euronext Securities Copenhagen, subsist.
- 24.4 For the avoidance of doubt, no Noteholder has the right to accelerate the Notes or otherwise request a prepayment or redemption of the Notes if a failure to admit the Notes to trading or maintain admission to trading of the Notes in accordance with Clauses 24.2 or 24.3 above occurs.

25. Governing law and jurisdiction, recognition of write-down and conversion powers

- 25.1 These Terms and Conditions and the Notes shall be governed by, and construed in accordance with, Danish law.
- 25.2 The City Court of Copenhagen (Da: *Københavns Byret*) shall have exclusive jurisdiction to settle any dispute arising from or connected with these Terms and Conditions and the Notes.
- 25.3 Recognition of write-down or conversion powers
 - (a) Notwithstanding and to the exclusion of any other term of the Notes or any other agreements, arrangements or understanding between the Issuer and any Noteholder, by its acquisition of the Notes, each Noteholder acknowledges and accepts that any liability arising under the Notes may be subject to (without limitation) the exercise of any Danish Statutory Loss Absorption Powers (including, for the avoidance of doubt, in accordance with Article 48 of the BRRD.
 - (b) Any failure or delay to give notice to the Noteholders on the exercise of any Danish Statutory Loss Absorption Powers shall not prejudice the validity and enforceability of the exercise of any Danish Statutory Loss Absorption Powers.

5. Amount and Use of Proceeds

5.1 An amount equal to the net proceeds from the Issuance of the Notes, being approximately DKK 198,600,000, have been and will be applied by the Issuer for general banking purposes, including, without limitation, asset/liability management, strategic liquidity management and/or to fulfil own funds requirements.

6. Description of BankNordik and the BankNordik Group

- 6.1 Introduction
- 6.1.1 History, business and market

BankNordik was established on the Faroe Islands in 1906 under the name Føroya Banki and merged with Sjóvinnubankin in 1994. The merged bank continued under the name Føroya Banki. In 2010 it changed the name to BankNordik.

BankNordik's vision is to be best in class to maintain existing customers and to attract new customers. The vision is to be accomplished through BankNordik's mission - to make it easy for its customers to make the right decision. BankNordik's current strategy is under the headline "A sustainable group in a digital society" with sound focus on contributing positively to sustainability and increasing the digitization of services delivered in order to improve the customer experience and being more efficient going forward.

The Group covers the Faroe Islands and the larger cities in Greenland, offering relevant and competitive financial products and services combined with competent advisory services for private individuals and enterprises.

BankNordik due to its relative size to the Faroese economy has been designated as a systematically important financial institution ("SIFI") on the Faroe Islands by the Danish FSA.

BankNordik Group is a relevant and strong financial institution offering a wide range of financial products and services to personal and corporate. BankNordik Group offers financial advice and support to the customers of BankNordik Group via a limited number of branches (five on the Faroe Islands and 1 in Greenland, Nuuk) and user-friendly digital self-services.

BankNordik Group's registered office and principal place of business is Oknarvegur 5, Thorshavn, Faroe Islands. Contact information for BankNordik's head office is: Telephone: +29833030 and BankNordik's main website is: www.banknordik.fo along with a local Greenlandic website www.banknordik.gl and an investor-related website www.banknordik.com. Information on BankNordik's websites does not form part of this Prospectus unless that information is incorporated by reference into this Prospectus (see "*Documents incorporated by reference*").

BankNordik is registered with the Faroese Business Authority under registration number 10 and in the Danish Business Authority as a limited liability company (in Danish: *aktieselskab*) and operates under Faroese legislation. BankNordik's Legal Entity Identifier (LEI) code is 5299005W3ZSR35ZYJH59. The legal name of the Bank is P/F BankNordik. BankNordik has the following registered secondary names: Amagerbanken, BankNordic, Foroya Bank, Foroya Banki, Føroya Banki, Invest Nordik, Nordic Bank, Nordic Banki, Nordik Bank, Nordik Banki, Nordik Íløgur, NordikInvest, Norðoya Banki, Pension Nordik, Pensjón Nordik, Sjóvinnubankin, Trygd Nordik.

The share capital amounts to DKK 192,000,000, divided into shares of nominally DKK 20 each. The share capital is fully paid up, and no shares carry special rights. There are no restrictions on the negotiability of the shares, but no shareholder may on its own behalf and/or as proxy holder for others cast a vote of more than 10% of the total votes.

The BankNordik share is listed on Nasdaq OMX, Copenhagen.

The five largest shareholders combined hold about 66.1 per cent of the share capital. The remaining 33.9 per cent of the share capital is divided between BankNordik's approximately other 8,400 shareholders.

BankNordik has one shareholder who holds more than 33.33 per cent of the share capital:

• The Government of the Faroe Islands 34.8 per cent of the shares, where the ownership percentage stated corresponds to what had been disclosed to BankNordik at the time of publication of the 2023 annual report.

The following shareholders have hold at least 5 per cent of the share capital:

- Ruth Holding ApS holds 14.6% of the shares.
- GTM Familie Holding ApS holds 10.4% of the shares.
- Kaldbakur, ehf holds, 5.1% of the shares.

where the ownership percentages stated corresponds to what had been disclosed to BankNordik at the time of publication of the 2023 annual report or in a major shareholder notification notified to BankNordik at a later date, if any.

6.1.2 Organisation and activities of the BankNordik Group

The BankNordik Group includes the following companies:

- P/F BankNordik (parent company)
- P/F Tryggingarfelagið Trygd ("Trygd") (subsidiary)
- P/F Lívstryggingarfelagið NordikLív ("NordikLív") (subsidiary)
- P/F Skyn ("Skyn") (subsidiary)

BankNordik Group's organisational structure is shown below:



6.1.3 BankNordik

BankNordik is a full-service bank for private individuals and small and medium-sized enterprises covering the Faroe Islands and the major cities in Greenland.

BankNordik's business strategy is based on the sound the values of being pro active, committed and competent with a solid focus on being able to make it easy for its customers to make the right financial choice. As an example, advisory services always take outset in the needs, values and dreams of the customer. This means that BankNordik's approach is to look at the full financial situation of the customer.

BankNordik works systematically and strategically to develop a dynamic and customer-oriented culture. BankNordik's culture is crucial in retaining and developing BankNordik's unique position relative both to customers and to being an attractive workplace for employees. BankNordik sees a clear link between strategy and culture.

6.1.4 Trygd

Trygd is BankNordik's non-life insurance company active only on the Faroe Islands serving both private and corporate customers' non-life insurance needs.

6.1.5 NordikLív

NordikLív is life insurance company only active on the Faroe Islands primarily focused on servicing private individuals with severe illness coverage and life insurances. All advisory and sales are made through BankNordik or Trygd.

6.1.6 Skyn

Skyn is a real estate broker only active on the Faroe Islands.

- 6.1.7 Strategy
- 6.1.7.1 BankNordik's strategy

Implementation of BankNordik's current strategy "A sustainable group in a digital society" is focused on the Group's three main stakeholders i) the employees, ii) the customers and iii) the shareholders along with due consideration to the fourth underlying stakeholder iv) the two societies, Faroe Islands and Greenland, the Group operates in.

The strategy has four pillars supplemented with some overall financial and ESG related targets.

The four pillars of the current strategy:

- Digital customer experiences: The customer shall be able to do all banking related business digital. However, customers shall always have the opportunity to opt for in-person advisory services
- Efficient process: Automation and efficient internal processes are a prerequisite for the Group to be able to deliver sound customer experiences along with being able to stay competitive going forward
- Well-being and development of competencies among the employees: Continuous development of the well being of the employees along with their competencies shall secure a good working environment and the best service to our customers

• Sustainability: Focus on taking due consideration to the local societies where we operate along with the environment and the employees, e.g. by reducing direct emissions of CO2 (Scope 1) and even split between the sexes.

And based on the execution of the above the long term financial and ESG related main targets are:

- Employee satisfaction 80 (equals a high score on a scale from 0-100)
- ROE >12% (after tax and with normalised losses))
- CO2 emissions 0 (Scope 1)

6.1.7.2 Trygd's strategy

To be a competitive non-life insurance partner to corporate customers and private individuals on the Faroe Islands.

6.1.7.3 NordikLív's strategy

NordikLív's primary focus is delivering severe illness coverage and life insurances under the Faroese pension schemes and almost all operating activities of the company are outsourced to one of the largest Danish providers of life insurances Forenede Gruppeliv or to BankNordik and Trygd.

6.1.7.4 Skyn's strategy

To deliver high quality real estate brokage services on the Faroese housing market along with maintaining the position as one the largest brokerage firms active on the Faroese housing market.

6.2 Business Segments

The BankNordik Group consists of three main business segments: Corporate customer activities, Private customer activities and Insurance activities.

Corporate and private customer activities are defined as:

- all business transactions/trading conducted with corporate/private customers, where the Group generates earnings in the form of interest-rate differentials, commissions, fees or additional brokerage fees;
- returns on equity investments relating to corporate/private customer activities; and
- returns on derivative transactions entered into in order to hedge a market risk on corporate/private customer activities.

Earnings and costs related to treasury and liquidity management and not included in the above business segments.

Insurance activities is defined as all activity in Trygd.

Selected information about the BankNordik Group's business segments as of the year 2023 is presented on page 61 of the BankNordik Group's 2023 annual report. See also "*Documents incorporated by reference*".

6.3 Associated companies

BankNordik Group has equity investments in the Faroese IT and hosting company Elektron P/F which is jointly owned by the two Faroese banks, the two Faroese saving banks and the Faroese Government. As of 31 December 2023, BankNordik Group held 34 per cent of the shares issued by Elektron P/F.

6.4 Business partners

The BankNordik Group is member of SDC (Skandinavisk Datacentral A/S). SDC is characterized by having all for Faroese banks/saving banks along with a large number of Danish, Norwegian and Swedish member banks which together with BankNordik, contribute to develop the capacity required for future banking operations of SDC member banks.

Other important business partners in various product and advisory service areas include:

- DLR Kredit A/S (mortgage-credit institution)
- BI Asset Management Fondsmæglerselskab A/S (investment associations)
- Mastercard (payment services)

6.5 Financial statement

The audited consolidated financial statements as at and for the years ended 31 December 2023 and 31 December 2022 are incorporated in this Prospectus by reference. See "*Documents incorporated by reference*".

6.6 Faroese and Greenlandic Banking Market

The Faroese and Greenlandic banking market is characterized by having low and stable number of banks being active in these two markets. In the Faroese market only two banks, servicing approximately 80% of the market evenly split between them, BankNordik being one of them, and two minor saving banks are active, while in the Greenlandic market only two domestic banks are active – BankNordik, approximately holding 20% of the market, and Grønlandsbanken holding the remaining 80%. In both markets foreign banks are active through cross boarder activities and before mentioned market shares are excluding the activities of foreign banks.

BankNordik's main competitor in the Faroese market is the Faroese owned Betri Banki and in Greenland it is the listed bank Grønlandsbanken.

6.7 The Board of Directors of BankNordik

The Board of Directors of BankNordik is composed of nine members, including six elected by the general meeting and three elected by the employees. The term of office for members elected by the general meeting is one year, and for the employee representatives four years.

As at the date of this Prospectus, BankNordik's Board of Directors consists of:

Birita Sandberg Samuelsen

Chairman of the Board of directors

Born: 1975

Gender: Female

Nationality: Faroese

Independent

Education

- 2007 Lawyer, admitted.
- 2004 Cand.jur, University of Copenhagen, Denmark.

Directorships and other offices

- Birita Sandberg Samuelsen was board member of the Financing Fund of 1992, hereof chairman for the period 2015-2017.
- Since 2020 Birita Sandberg Samuelsen has been chairman of the board at P/F Vága Floghavn. For the period 2014-2017.
- Since 2016 Birita Sandberg Samuelsen has been lawyer and partner at the law firm Sókn Advokatar.
- For the period, 2010-2016 Birita Sandberg Samuelsen was lawyer and partner at the law firm.
- Advokatskrivstovan Í/F.
- Birita Sandberg Samuelsen was for the period 2007-2011 adviser at the Faroese Maritime.
- Authority (Sjóvinnustýrið), while she for the period 2004-2007 was principal at the Faroese Parliament.
- For the period 2001-2002 Birita Sandberg Samuelsen was student at the Danish Labour.
- Market Insurance, and for the period 1997-1999 student at The Prime Minister's Office in Denmark.

Rúni Vang Poulsen

Deputy-chairman of the Board of Directors

Born: 1975

Gender: Male

Nationality: Faroese

Independent

Education

- 2013 Orchestrating Winning Performance IMD Lausanne, Switzerland.
- 2003 2006 MSc Business Administration and Auditing, Copenhagen Business School, Denmark.
- 1998 2002 HD (Graduate Diploma in Business Administration), Aarhus Business School, Denmark.

Management positions

- For the period 2010-2021 Rúni Vang Poulsen was Chief Executive Officer of Smyril Line P/F.
- For the period 1997-2006 Rúni Vang Poulsen was Senior Advisor at SPEKT, audit company.
- Rúni Vang Poulsen is former vice-chairman of the board of P/F BankNordik and former board member of the Financing Fund of 1992.

Birger Durhuus

Member the Board of Directors

Born: 1963

Gender: Male

Nationality: Faroese

Independent

Education

• 1990 Master of Finance, Copenhagen Business School.

Directorships and other offices

- In 2023 Birger Durhuus joined Absolute Return Partners as Partner.
- For the period, 2017-2023 Birger Durhuus was CEO of Asgard Asset Management, Copenhagen.
- For the period, 2015-2017 Birger Durhuus was advisor at Moma Advisors A/S.

- Birger Durhuus was for the period 2007-2015 Head of External Solutions & Risk Management at Danske Capital.
- For the period 1999-2007 Birger Durhuus was Head of Sales & Research at Nordea Markets.
- For the period 1993-1999 he was Chief Analyst at Nykredit Markets.
- For the period 2009-2015 and the period 2017-2018 Birger Durhuus was chairman of the board of Atlantic Petroleum.

Marjun Eystberg

Member of the Board of Directors

Born: 1963

Gender: Female

Nationality: Faroese

Independent

Education

- 2019 Process leader, Copenhagen Coaching Center.
- 2011 Master in communication, Probana 2009 HR Master, Probana.
- 2005-2006 Master in management, Financial Sector Education Center.
- 1989-1992 Organisation and Strategy, Copenhagen Business School.

Directorships and other offices

- Since 2019 Marjun Eystberg has been HR Manager at Varðin Pelagic.
- Furthermore, Marjun Eystberg since 2012 has been manager at INPUT, primarily providing consultancy within management and HR.
- For the period 2003-20210 Marjun Eystberg was employed at Betri Banki as Executive Area Manager and Manager of Personnel Department.
- Marjun Eystberg is board member of the Faroese Business Development Fund (Framtak), and for the period 2017-2020 she was board member of the Financing fund of 1992.

Kristian Reinert Davidsen

Member the Board of Directors

Born: 1966

Gender: Male

Nationality: Faroese

Independent

Education

- 2002 HD (Graduate Diploma in organization and management).
- Copenhagen Business School.
- 1991 MSc Engineering, E-division, DTU, Denmark.
- 1989 Academy Engineer, ET-division, DIA, Denmark.

Directorships and other offices

- Since 2015 Kristian Reinert Davidsen has been CEO of Tusass (TELE Greenland).
- For the period 2006-2015 Kristian Reinert Davidsen was CEO of Faroese Telecom.
- For the period 2003-2006 he was managing director of the Faroese public transport company Strandfaraskip Landsins.
- From 1996 to 2003 Kristian Reinert Davidsen was employed at Ericsson in Denmark, working as department manager for Support Services and squad leader for Network Design.
- Kristian Reinert Davidsen is chairman of the board of LBF (Engineering consultancy) and board member of the Telecommunication and IT company P/F NEMA.

Tom Ahrenst

Member of the Board of Directors

Born: 1960

Gender: Male

Nationality: Danish

Independent

Education:

- 2012 Executive Management Program, Columbia Business School.
- 1992 Executive Management Program, Wharton Business School.
- 1985 HD Accounting, Copenhagen Business School.

Directorships and other offices:

- Since 2023 Tom Ahrenst has been senior advisor, EVP for Nykredit.
- For the period 2000-2022 Tom Ahrenst was Executive Vice President at Nykredit, and Senior Vice President from 1997-1999.
- For the period 1978-1997 Tom Ahrenst held various positions in Danske Bank, including Vice President in the Corporate Credit Department.
- Tom Ahrenst is a board member of Core Property Management P/S.

Alexandur Johansen

Member of the Board of Directors

Born 1979

Gender: Male

Nationality: Faroese

Elected by the employees for the term 2022-2026

Education:

• Financial education and subsequent continuing education within financial and insurance aspects.

Principal occupation:

• P/F Trygd - Commercial Insurance - leader of corporate and private department.

Directorships and other offices: None

Kenneth Samuelsen

Member of the Board of Directors

Born: 1966

Gender: Male

Nationality: Faroese

Elected by the employees for the term 2022-2026

Education:

• Financial education.

Principal occupation:

• BankNordik - IT department.

Directorships and other offices: None

Rúna Hentze

Member of the Board of Directors

Born: 1966

Gender: Female

Nationality: Faroese

Elected by the employees for the term 2022-2026.

Education:

• Financial education supplemented with different banking related courses.

Principal occupation:

• BankNordik - Back Office.

Directorships and other offices: None

6.8 The Executive Management of BankNordik

Turið F. Arge

CEO

Born: 1982

Gender: Female

Nationality: Faroese

Year of joining the Executive Management: 2022

Education:

- MSc in Business Economics and Auditing, Aarhus Business School.
- Executive MBA, Henley Business School.

Principal occupation:

• CEO at P/F BankNordik.

Board positions:

• P/F Trygd, P/F NordikLív, P/F Skyn and the Faroese Banking organisation

The business address of the members of BankNordik's Board of Directors and Executive Management is Oknarvegur 5, 100 Thorshavn, Faroe Islands.

6.9 Statement on conflicts of interest

There are no potential conflicts of interests between any duties to BankNordik of members of the Board of Directors or the Executive Management and their private interests and/or other duties.

- 6.10 Capital and solvency need
- 6.10.1 SIFI Status

Due to the Group's banking activities relative size compared to the Faroese economy, BankNordik is appointed as and treated as an SIFI status, including required to include a SIFI-buffer in its capital buffer requirement at the time of this Prospectus of 2 per. cent.

6.10.2 Capital management

Capital management is based on the EU Capital Requirements Regulation ("CRR"), the EU Capital Requirements Directive ("CRD") and the EU Recovery and Resolution Directive ("BRRD") all put in to effect by law and/Royal Decree on the Faroe Islands. As part of capital management, the Group has drawn up a capital plan to ensure that the Group has sufficient capital to comply with current legislation and at all times meet its own solvency targets. The legislation concerns:

- Calculation of capital, risk exposures and capital requirements.
- Calculation of individual solvency need.
- Disclosure requirements.

The Group's capital plan has been supplemented by a recovery plan comprising a number of relevant risk and capital indicators for the Group with associated limit values, stress test scenarios and recovery measures to ensure that the Group is able to identify problems in time and implement measures to ensure the viability of the Group. The Group regularly monitors developments in risk indicators.

Based on legal requirements and the limit values for capital indicators set in the recovery plan, the Group has set an external capital target of a CET1-ratio of 20 per cent and total capital target of 2 percentage points above the regulatory requirement, covering the solvency need plus the capital conservation buffer and the SIFI capital buffer, as well as an additional excess cover. The capital target corresponds to the yellow light indicator in the recovery plan and ensures that the Group can absorb future capital requirements, e.g. in the form of increased buffer requirements.

With a capital ratio of 29.4 per cent as of 31 December 2023, the Group thus has an excess cover of 9.4 percentage points. Relative to the statutory requirement of 10.3 per cent, the excess cover is 19.0 percentage points.

The Group's capital requirement comprises as of the date of this Prospectus a solvency need of 10.3 per cent and the combined capital buffer requirement of 4.81 per cent (exclusive of the SIFI buffer of 2.0 per cent), which currently consists of the capital conservation buffer of 2.5 per cent, the systemic buffer of 1.49 per cent, and the countercyclical capital buffer of 0.82 per cent. The systemic buffer along with the countercyclical buffer are alone effective in respect of Faroese exposures, leaving e.g. out the Greenland exposures. The Faroese exposures is approximately 78% of the Group's total exposures as of the date of this Prospectus

The long-term capital planning of the Group is based on economic projections taking account of the Group's vision, but under different macro-economic and idiosyncratic stress scenarios and recognising the anticipated effects of future legislation.

6.10.3 Own Funds

Overall, Group's own funds consist of Common Equity Tier 1 capital, Additional Tier 1 capital and Tier 2 capital. Own funds are calculated with a view to calculating capital ratios that help to express the Group's capital excess cover in relation to the Group's targets and regulatory capital requirements.

The audited consolidated financial statements as at and for the years ended 31 December 2022 and 31 December 2023 are incorporated in this Prospectus by reference. See "*Documents incorporated by reference*", including the specification of own funds per 31 December 2022 and 2023 in said documents.

Further the Group's unaudited Risk Management Report 2023 is incorporated in this Prospectus by reference. See "*Documents incorporated by reference*".

6.10.4 Risk exposure amount and capital ratios

The total risk exposure amount (REA) is used to set the minimum capital requirement and to calculate the Group's capital ratios, capital buffers, individual solvency needs and MREL requirement. This risk exposure represents the basis for determining the capital that must be reserved relative to the risk undertaken by the BankNordik Group involving credit, market, and operational risks.

The BankNordik Group uses the following methods for calculating the risk exposure:

- The standardised approach for calculating credit risk.
- The standardised approach for calculating market risk.
- The market value method to calculate counterparty risk.
- The basic indicator approach to calculate operational risk.
- The standardised approach for calculating the CVA risk.

The Group's CET1 capital ratio target has been calculated at 20 per cent with a CET1 capital ratio of 25.8 per cent as of 31 December 2023, the Group thus has an excess cover of 5.8 percentage points. Relative to the statutory requirement of 10.3 per cent, the excess cover is 19.0 percentage points.

6.10.5 Individual solvency requirement

The Danish Financial Business Act, the CRD and SREP guidelines determine the requirements for the individual solvency requirement and encompasses any additional capital requirements to cover risks which are not adequately covered by the minimum own funds requirement of 8 per cent according to CRR.

BankNordik uses the 8+ model which is based on a minimum requirement of 8 per cent of the total risk exposure (pillar 1 requirement). Normal risks are assumed to be covered by the 8 per cent capital requirement. Furthermore, the BankNordik assesses the extent to which additional capital is needed to cover risk areas not included in the 8 per cent requirement (pillar 2). The total capital need is obtained by adding together the capital need according to pillar 1 and pillar 2.

The model is based on the "Guidelines on adequate own funds and solvency need for credit institutions" from the Danish FSA, which determines benchmarks and methods of calculation for any pillar 2 supplements within the following risk areas: Earnings, growth in lending, credit risk, market risk, liquidity risk, operational risk, leverage and regulatory maturity of capital instruments. Solvency need is calculated as the total capital need as a percentage of total risk exposure in accordance with the provisions of the CRR.

The Board of Directors of BankNordik determines adequate own funds and the individual solvency need for the Group. The Group assesses that the calculation of the total capital need is sufficient to cover the risks assumed by the Group.

A table showing the capital need per 31 December 2022 and 31 December 2023 in the financial statements for 2022 and 2023 respectively and all incorporated in this Prospectus by reference. See *"Documents incorporated by reference"*.

6.10.6 Combined buffer requirement

As a result of implementation of CRD in the Danish Financial Business Act as in force on the Faroe Islands, the Group is obliged to comply with the combined capital buffer requirement, which for the Group consists of a capital conservation buffer, the countercyclical capital buffer, the systemic buffer and the SIFI-buffer. The capital buffer requirement can only be met through Common Equity Tier 1 capital. Non-compliance with the capital buffer requirement will result in restrictions on the Group's possibilities to pay dividends and make other distributions.

With the designation as a SIFI, the Group must meet a SIFI buffer requirement of 2 per cent of the total risk exposures. The SIFI buffer is set on the basis of the Group's systemic importance and must be met through Common Equity Tier 1 capital.

The capital conservation buffer helps to ensure a more robust financial sector in terms of capital and constitutes a fixed capital requirement of 2.5 per cent of the total risk exposure. The countercyclical capital buffer along with the systemic buffer is set by the Minister for Industry, Business and Financial Affairs after recommendations from The Systemic Risk Council. In December 2021, the Minister for Industry, Business and Financial Affairs decided that the systemic buffer is set at 2.0 per cent of BankNordik's Faroese exposures and countercyclical capital buffer was activated at a rate of 1.0 per cent of the Faroese exposures from March 2023. No systemic buffer of countercyclical capital buffer is currently in place or expected to be put in place in respect of the Group's Greenlandic exposures.

At the end of 2023, the combined capital buffer requirement consisted of the capital conservation buffer (2.5 per cent), the SIFI buffer (2.0 per cent) and the systemic buffer (2 per cent) and was calculated at DKK 464 million.

6.10.7 Excess cover in relation to the total capital requirement

The Group's capital ratio amounted to 29.4 per cent at the end of 2023, corresponding to an excess cover of 12.4 percentage points relative to the total capital requirement of 17.1 per cent, comprising the solvency requirement of 8.0 per cent, the supplementary solvency need of 2.3 per cent and the combined capital buffer requirement of 6.8 per cent. The solvency requirement of 8.0 per cent must be covered by at least 4.5 per cent Common Equity Tier 1 capital, while at least 6.0 per cent of the solvency requirement must be covered by Tier 1 capital. Tier 2 capital may account for up to 2.0 per cent. The supplementary solvency need and the combined capital buffer requirement can only be covered by Common Equity Tier 1 capital.

At the end of 2023, the Group had excess cover of DKK 833 million relative to the Common Equity Tier 1 capital requirement, and excess cover of DKK 833 million relative to the Tier 1 capital requirement. The excess cover for the Group's total capital requirement is DKK 844 million.

6.10.8 MREL requirement

According to the Danish Financial Business Act, the Danish FSA and Finansiel Stabilitet are responsible for preparing resolution plans for banks in distress. These plans state that a minimum requirement for own funds and eligible liabilities (MREL) must be determined for the individual bank. The MREL will ensure that a bank in distress has sufficient eligible liabilities to cover losses in the bank and to recapitalise the bank so that critical functions can be continued without using public funds. For SIFIs, an individual resolution strategy has been drawn up. The overall aim of the strategy is to ensure that the Group can be returned to the market as a viable financial institution after restructuring. This will be through recapitalisation of the resolution group on a consolidated basis at resolution group level by writing down and converting creditors' claims.

There is no minimum EU-wide level of the MREL Requirement – each resolution authority is required to make a separate determination of the appropriate MREL Requirement for each resolution group within its jurisdiction, depending on the resolvability, risk profile, systemic importance and other characteristics of each credit institution. The resolution strategy for the BankNordik Group is single point of entry at the level of the Issuer being the resolution entity and the Issuer and its subsidiaries being the resolution group.

The designation as a SIFI means that the Group will be subject to an MREL Requirement corresponding to the sum of two times the solvency need, the combined buffer Requirement, the capital conservation buffer and the SIFI buffer.

In November 2023, the Danish FSA published the latest MREL Requirement for BankNordik to be phased in over a period running until 1 June 2026 (the "2023 MREL Requirement Decision"). As of 1 June 2024, BankNordik on a consolidated basis must fulfil an MREL Requirement on 21.7 per cent of BankNordik's total REA. When BankNordik's MREL Requirement has been fully phased in by 1 June 2026, the MREL Requirement for BankNordik will be 25.5 per cent of BankNordik's total REA.

The MREL Requirement will be set at the end of each year on the basis of numbers for that preceding year (but may be updated over the year and may change over the year and the contemplated phase-

in of BankNordik's MREL Requirement may also deviate from what is set out and anticipated in the 2023 MREL Requirement Decision.

The MREL Requirement can be met by, inter alia, Common Equity Tier 1 instruments, Additional Tier 1 Capital instruments, Tier 2 Capital instruments and MREL eligible liabilities, including non-preferred senior instruments and senior preferred instruments with a maturity of more than 1 year. Currently, the Group expects that the requirement for the MREL add-on will be met by issuing non-preferred senior instruments and senior preferred instruments, such as the Notes. Up to the final phase-in of the MREL add-on on 1 June 2026, the Group expects to have to issue a minimum of DKK 650 million in non-preferred senior instruments and a minimum of DKK 330 million in senior preferred instruments, such as the Notes, to cover the MREL requirement, depending on changes in risk exposures, solvency need and requirements for the countercyclical capital buffer.

6.10.9 Future capital adequacy rules

On 27 October 2021, the European Commission announced a proposal to amend the CRR and the CRD, also known as Basel IV. The proposal entails that the expected date of entry into force of the amendments will be pushed from 1 January 2023 to 1 January 2025.

The Group uses the standardized approach for calculating risks and has noted that changes to credit risk and market risk have been proposed under the standardized approach, including stricter capital requirements for unutilized credit facilities and exposures secured by mortgages in real property. With current capitalization the Group expect that also after phasing in Basel IV, the Group will meet its capital target.

6.10.10 Leverage Ratio

The leverage ratio is calculated as Tier 1 capital according to the fully phased-in definition, in relation to the total exposure. Through its leverage policy, the Group has set a framework for managing and monitoring the risk of excessive leverage. The aim of the Group's leverage is set in accordance with the Group's risk strategy, and the Group assesses that a leverage ratio of more than 8.0 per cent is appropriate in relation to the Group's business model..

Leverage risk is defined in the CRR, which stipulates that a bank must have a leverage ratio of at least 3.0 per cent.

At the end of 2023, the Group had a leverage ratio of 13.8 per cent and thus meets both the minimum requirement and its leverage target by a solid margin.

6.11 Liquidity

Liquidity Risk is the risk that a lack of funding leads to excessive costs or prevents the BankNordik Group from maintaining its business model or fulfilling its payment obligations.

Liquidity Risk is measured and controlled through a range of metrics with applicable limits, including net stable funding ratio (NSFR), liquidity covered ratio (LCR) and LCR-ratio under stress scenarios.

The Group's operations are predominately funded through the following two material funding sources:

• Deposits (retail and corporate deposits).

- Senior funding and MREL eligible debt.
- Shareholders' equity and subordinated debt.

Of deposits and other payables as at 31 December 2023 totalling DKK 8,702 mill, DKK 4,732 mill. is covered by the Danish Deposit Guarantee Scheme. The remaining deposits of approximately DKK 3,970 mill. are not covered by a guarantee scheme.

For short-term liquidity in foreign currency, the Group uses money-market funding from other credit institutions.

The below table illustrates the maturity structure of the Group's total funding as of 31 December 2023.

Туре	Currency	Nominal value	Booked value (DKK)	Maturity Date	First Call Date
AT1	DKK	150.000.000	151.531.838	N/A	30-09-2024
T2	DKK	150.000.000	99.650.000	24-06-2031	24-06-2026
SNP	DKK	150.000.000	149.578.125	18-06-2026	18-06-2026
SNP	SEK	300.000.000	199.819.686	31-03-2027	31-03-2026
SNP	DKK	250.000.000	248.183.854	02-12-2030	01-12-2027
SP	DKK	125.000.000	125.000.000	01-11-2024	01-11-2024
SP	DKK	250.000.000	250.000.000	01-11-2025	01-11-2025
SP	DKK	190.000.000	189.743.333	10-03-2025	10-03-2025
SP	DKK	300.000.000	285.714.286	08-10-2028	08-10-2028
SP	DKK	200.000.000	198.808.854	22-11-2028	22-11-2027

In the coming years, the issuance of NPS instruments will also be a material funding source for the Group.

6.11.1 Net stable funding ratio

The net stable funding ratio (NSFR) focuses on the long-term funding and serves to define the minimum acceptable amount of stable funding, based on the Group's liquidity characteristics of assets and activities over a one-year time horizon. The NSFR-assumptions are defined in CRR.

As of end 2023, the Group's NSFR-ratio was 152 per cent.

6.11.2 Liquidity covered ratio

The Group is required to maintain an adequate level of unencumbered high-quality liquid assets that can be converted into cash to meet liquidity needs over a 30-calendar day horizon under a predefined significantly severe liquidity stress scenario. This is measured by the Liquidity Coverage Ratio (LCR) which is based on a 30-day liquidity stress scenario, with assumptions defined in the LCR regulation.

The primary focus in the management of the Group's LCR buffer is on the total amount of LCR eligible Level 1 and Level 2 assets whereas the split between Level 1a and other eligible LCR assets is of secondary importance as far as overall compliance is achieved.

Relative to the LCR Regulation, the Group calculated its LCR as of 31 December 2023 at 228.2 per cent compared to the Group's minimum target for LCR of 150 per cent.

6.12 Sustainability

Sustainability has entered the strategic agenda for the financial sector in earnest. The sector has a vital role in the transition to a green economy, and it is encountering more regulation and increasing reporting requirements.

BankNordik embraces this movement and has incorporated ESG initiatives and targets in its current strategy, including

- A target of 0 for direct CO2 emissions (Scope 1) in 2025
- A reduction in indirect CO2 emissions (Scope 2) from 240 tonnes in 2023 to 170 tonnes in 2025
- A gender diversity not only on management and board level, but also per each department/unit in the range of 40-60 per cent.

6.13 Credit rating

BankNordik is rated by Moody's. On 20 November 2023, Moody's assigned a long-term unsecured rating of A1 and a short term unsecured rating of P-1 to BankNordik with a positive outlook. These ratings are based on a baseline credit assessment (BCA) of baa1. The rating is attached to BankNordik's company announcement no. 23/2023 which can be found through <u>this link</u>.)

6.14 Legal and arbitration proceedings

The BankNordik Group is and may become involved in various disputes and legal proceedings on the Faroe Islands, in Greenland and other jurisdictions, including litigation and regulatory investigations.

Provisions for legal disputes are recognised where a legal or constructive obligation has incurred as a result of past events, and it is probable that there will be an outflow of resources that can be reliably estimated. The BankNordik Group estimates legal disputes on the basis of an evaluation of the most likely outcome. Provisions are measured at the present value of the anticipated expenditure for settlement of the legal or constructive obligation that reflects the risks specific to the obligation.

As at the date of this Prospectus there are no governmental, legal or arbitration proceedings (including any such proceedings which are pending or threatened of which the Issuer is aware) which may have, or have had in the past 12 months, significant effects on the financial position or profitability of the Issuer and/or the BankNordik Group, and there have not been any such proceedings within the past 12 months.

7. General Information

- 7.1 The Issuer has obtained all necessary consents, approvals and authorizations in Denmark in connection with the issue and performance of the Notes. The issuance of Notes was duly authorized by a resolution of the Board of Directors of the Issuer passed on 25 September 2023.
- 7.2 There has been no significant change in the financial performance or position of the Issuer or the Group since 31 December 2023, nor has there been any material adverse change in the financial position or prospects of the Issuer since the date of its last published audited financial statements.
- 7.3 There are no governmental, legal or arbitration proceedings (including any such proceedings which are pending or threatened of which the Issuer is aware) as at the date of this Prospectus which may have, or have had in the past 12 months, significant effects on the financial position or profitability of the Issuer and/or the Group, and there have not been any such proceedings within the past 12 months.
- 7.4 The International Securities Identification Number ("ISIN") for the Notes is DK DK0030529664. The common code for the Notes is 27263964 The Financial Instrument Short Name ("FISN") for the Notes is P F BankNordi/6.20/ PrfSnr 2028. Classification of Financial Instruments Code ("CFI") for the Notes is DTVUGB.
- 7.5 There are no material contracts entered into in the ordinary course of the Issuer's business which could result in any member of the Group being under an obligation or entitlement that is material to the Issuer's ability to meet its obligations to Noteholders in respect of the Notes.
- 7.6 Where information in this Prospectus has been sourced from third parties, this information has been accurately reproduced and, as far as the Issuer is aware and is able to ascertain from the information published by such third parties, no facts have been omitted which could render the reproduced information inaccurate or misleading. The source of third-party information, such as the Faroese Economic Council, is identified where used.
- 7.7 Copies of the following documents (and an English translation of the Articles of Association) will, when published, be available as follows:
 - the Articles of Association of the Issuer (available at <u>https://www.banknordik.com/organisation/governance/articles-of-association</u>);
 - (ii) the memorandum of incorporation of the Issuer (available at https://www.banknordik.com/share-data/debt-and-funding); and
 - (iii) a copy of this Prospectus (available at <u>https://www.banknordik.com/share-data/debt-and-funding</u>).

The document specified in sub-paragraph (i) is a direct English translation of the Faroese language original. In the event that there are any inconsistencies or discrepancies between the Faroese language version and the English translation thereof, the original Faroese language version shall prevail.

7.8 PriceWaterhouseCoopers Statsautoriseret Revisionpartnerselskab, CVR no. 33 77 12 31, represented by state authorised public accountant Benny Voss (mne15009) with its registered address at Strandvejen 44, 2900 Hellerup, Denmark, and Januar Løggilt Grannskoðanarvirki P/F, business registration no. 5821, represented by Fróði Sivertsen (mne32257) with its registered

address at Óðinshædd 13, Postrúm 30, 110 Tórshavn, have audited the Issuer's and the Group's accounts, which were prepared in accordance with the International Financial Reporting Standards (IFRS), without qualification, for the financial years ended 2022 and 2023.

- 7.9 The Notes were issued by the Issuer and subscribed by the Arranger on the Issue Date in accordance with an issuance agreement (the "Issuance Agreement") dated 20 November 2023. The Arranger has sold the Notes to certain qualified investors. The Issuer has paid certain fees to the Arranger and reimbursed the Arranger for certain expenses incurred in connection with the issuance of the Notes. Furthermore, the Arranger and its affiliates has engaged in, and may in the future engage in, investment banking and other commercial dealings in the ordinary course of business with the Issuer or its affiliates. They have received, or may in the future receive, customary fees and commissions for these transactions. In addition, in the ordinary course of business of its activities, the Arranger and its affiliates may make or hold a broad array of investments and actively trade debt and equity securities (or related derivative securities) and financial instruments (including bank loans) for their own accounts and for the accounts of their customers. Such investments and securities activities may involve securities and/or instruments of the Issuer or the Issuer's affiliates. The Arranger and its affiliates may also make investment recommendations and/or publish or express independent research views in respect of such securities or financial instruments and may hold, or recommend to clients that they acquire, long and/or short positions in such securities and instruments.
- 7.10 The total expenses related to the admission to trading of the Notes are estimated by the Issuer to be approximately DKK 450,000.
- 7.11 The language of this Prospectus is English. Certain legislative references and technical terms have been cited in their original language in order that the correct technical meaning may be ascribed to them under applicable law.
- 7.12 In this Prospectus, references to websites or uniform resource locators (each a "**URL**") are inactive textual references and are included for information purposes only. The contents of any such website or URL shall not form part of, or be deemed to be incorporated into, this Prospectus unless that information is explicitly incorporated by reference into this Prospectus.

REGISTERED AND HEAD OFFICE OF THE ISSUER

P/F BankNordik

Oknarvegur 5 FO-110 Tórshavn Faroe Islands

ARRANGER

Nordea Bank Abp

Satamaradankatu 5 FI-00020 Nordea, Helsinki Finland

LEGAL ADVISER TO THE ISSUER

as to Danish law

Bech-Bruun Law Firm P/S

Gdanskgade 18 DK-2150 Nordhavn Denmark